%% if you are submitting an initial manuscript then you should have submission as an option here
%% if you are submitting a revised manuscript then you should have revision as an option here
%% otherwise options taken by the article class will be accepted
\documentclass[final]{FPSAC2022}
%% but DO NOT pass any options (or change anything else anywhere) which alters page size / layout / font size etc

%% note that the class file already loads {amsmath, amsthm, amssymb}

\newtheorem{thm}{Theorem}
%%% Commented out since we do this later, numbering with theorems. 
%\newtheorem{lem}{Lemma}

%\usepackage{lipsum}


%% define your title in the usual way
\title{Rowmotion on Fences}

%% define your authors in the usual way
%% use \addressmark{1}, \addressmark{2} etc for the institutions, and use \thanks{} for contact details

\author[S.~Elizalde, M.~Plante, T.~Roby, and B.~E.~Sagan]{Sergi Elizalde\addressmark{1}, Matthew Plante\addressmark{2}, Tom Roby\thanks{\href{mailto:tom.roby@uconn.edu}{tom.roby@uconn.edu}}\addressmark{2}, \and Bruce E. Sagan\addressmark{3}}

%% then use \addressmark to match authors to institutions here
\address{
\addressmark{1}Department of Mathematics, Dartmouth College \\ \addressmark{2}Department of Mathematics, University of Connecticut\\  
\addressmark{3}Department of Mathematics, Michigan State University\\ 
}

%% put the date of submission here
\received{\today}

%% leave this blank until submitting a revised version
%\revised{}

%% put your English abstract here, or comment this out if you don't have one yet
%% please don't use custom commands in your abstract / resume, as these will be displayed online
%% likewise for citations -- please don't use \cite, and instead write out your citation as something like (author year)
\abstract{A fence is a poset with elements $F=\{x_1,x_2,\ldots,x_n\}$ and covers
\vspace{-5pt}  % TCS
\[
x_1\lt x_2 \lt \cdots \lt x_a \gt x_{a+1} \gt \cdots \gt x_b \lt x_{b+1} \lt \cdots,
\vspace{-5pt}  % TCS
\]
where $a,b,\ldots$ are positive integers.  We investigate rowmotion on antichains and ideals
of $F$.  In particular, we show that orbits of antichains can be visualized using tilings.
This permits us to prove various homomesy results for the number of elements of an antichain
or ideal in an orbit.  Rowmotion on fences also exhibits a new phenomenon, which we call
homometry, where the value of a statistic is constant on orbits of the same size.  Along the
way, we prove a homomesy result for all self-dual posets and show that any two Coxeter
elements in certain toggle groups behave similarly with respect to homomesies which are
linear combinations of ideal indicator functions.  We end with some conjectures and avenues
for future research.}

%% put your French abstract here, or comment this out if you don't have one
%%\resume{\lipsum[2]}

%% put your keywords here, or comment this out if you don't have them yet
\keywords{fence posets, rowmotion, homomesy, homometry, dynamical algebraic combinatorics}

%% you can include your bibliography however you want, but using an external .bib file is STRONGLY RECOMMENDED and will make the editor's life much easier
%% regardless of how you do it, please use numerical citations, ie. [xx, yy] in the text

%% this sample uses biblatex, which (among other things) takes care of URLs in a more flexible way than bibtex
%% but you can use bibtex if you want
\usepackage[backend=bibtex]{biblatex}
%%\addbibresource{sample.bib}  %%% REPLACED THIS WITH THE ONE BELOW
\addbibresource{rfrefFPSAC.bib}  
%% note the \printbibliography command at the end of the file which goes with these biblatex commands

%%%% ABOVE IS STANDARD FPSAC; BELOW IS OUR ADDITIONS

\usepackage{amsfonts,latexsym,tikz,tikz-cd,wrapfig}%,hyperref,float}
%\usepackage[notref, notcite]{showkeys}

%%%Use this to get rid of proofs without having to remove them.
\usepackage{comment}% http://ctan.org/pkg/comment
%\excludecomment{proof}

%\newtheorem{thm}{Theorem}[section]
\newtheorem{prop}[thm]{Proposition}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{conj}[thm]{Conjecture}
\newtheorem{question}[thm]{Question}
\newtheorem{observation}[thm]{Observation}
\theoremstyle{definition}
\newtheorem{definition}[thm]{Definition}
\theoremstyle{remark}
\newtheorem{exa}[thm]{Example}

\newcommand{\uF}{\breve{F}}
\newcommand{\uS}{\breve{S}}
\newcommand{\cU}{{\mathcal U}}
\newcommand{\na}{\nabla}
\newcommand{\rhoh}{\widehat{\rho}}
\newcommand{\tauh}{\widehat{\tau}}
\newcommand{\cTh}{\widehat{\cT}}
\newcommand{\chih}{\widehat{\chi}}
\newcommand{\chib}{\ol{\chi}}
\newcommand{\cOb}{\ol{\cO}}
\newcommand{\bba}{\ol{b}}

%%%%%% For tikz figures:
\newcommand{\re}[1]{\filldraw (current) rectangle ++ (#1-.3,.7) ++(.3,-.7) coordinate (current);}
\newcommand{\ro}{\filldraw[fill=yellow] (current) rectangle ++(.7,.7)  ++(.3,-.7)
coordinate (current);}
\newcommand{\twobyone}{\filldraw[fill=red] (current) ++(0,.7) rectangle ++(.7,-1.7) ++(.3,1) coordinate (current);}
\newcommand{\blank}{\draw (current) ++(1,0) coordinate (current);}
%\newcommand{\releft}[1]{++(-.15,0) rectangle ++(#1-.15,.7) ++(.3,-.7)}
%\newcommand{\reright}[1]{rectangle ++(#1-.15,.7) ++(.15,-.7)}
\newcommand{\releft}[1]{ \filldraw (current) -- ++(-.3,0.23) -- ++(.3,.24) -- ++(-.3,.23) -- ++(#1,0) -- ++(0,-.7) ++(.3,0) coordinate (current);}
\newcommand{\reright}[1]{ \filldraw (current) -- ++(0,.7)-- ++(#1-.3,0) -- ++(.3,-.23) -- ++(-.3,-.24) -- ++(.3,-.23);}
\newcommand{\firstrow}{\coordinate (current) at (.15,-.85);}
\newcommand{\secondrow}{\coordinate (current) at (.15,-1.85);}
\newcommand{\thirdrow}{\coordinate (current) at (.15,-2.85);}
\newcommand{\fourthrow}{\coordinate (current) at (.15,-3.85);}
%%%%%



%%%% Macros for comments, to be removed at the end:
\newcommand{\sergi}[1]{\mbox{}{\sf\color{blue}[Sergi says: #1]}\marginpar{\color{blue}\Large$*$}} 
\newcommand{\bruce}[1]{\mbox{}{\sf\color{red}[Bruce says: #1]}\marginpar{\color{red}\Large$*$}} \newcommand{\trcomm}[1]{\mbox{}{\sf\color{OliveGreen}[Tom says: #1]}\marginpar{\color{OliveGreen}\Large$*$}} \newcommand{\matt}[1]{\mbox{}{\sf\color{orange!90!white}[Matt says: #1]}\marginpar{\color{orange!90!white}\Large$*$}} 


\newcommand{\ben}{\begin{enumerate}}
\newcommand{\een}{\end{enumerate}}
\newcommand{\ble}{\begin{lem}}
\newcommand{\ele}{\end{lem}}
\newcommand{\bth}{\begin{thm}}
\renewcommand{\eth}{\end{thm}}
\newcommand{\bpr}{\begin{prop}}
\newcommand{\epr}{\end{prop}}
\newcommand{\bco}{\begin{cor}}
\newcommand{\eco}{\end{cor}}
\newcommand{\bcon}{\begin{conj}}
\newcommand{\econ}{\end{conj}}
\newcommand{\bde}{\begin{defn}}
\newcommand{\ede}{\end{defn}}
\newcommand{\bex}{\begin{exa}}
\newcommand{\eex}{\end{exa}}
\newcommand{\overliner}{\begin{array}}
\newcommand{\earr}{\end{array}}
\newcommand{\btab}{\begin{tabular}}
\newcommand{\etab}{\end{tabular}}
%\newcommand{\beq}{\begin{equation}}
%\newcommand{\eeq}{\end{equation}}
%\newcommand{\bal}{\begin{align*}}
\newcommand{\bce}{\begin{center}}
\newcommand{\ece}{\end{center}}
\newcommand{\bpi}{\begin{picture}}
\newcommand{\epi}{\end{picture}}
\newcommand{\bpp}{\begin{picture}}
\newcommand{\epp}{\end{picture}}
\newcommand{\bfi}{\begin{figure} \begin{center}}
\newcommand{\efi}{\end{center} \end{figure}}
\newcommand{\bprf}{\begin{proof}}
\newcommand{\eprf}{\end{proof}\medskip}
\newcommand{\capt}{\caption}
\newcommand{\bsl}{\begin{slide}{}}
\newcommand{\esl}{\end{slide}}
\newcommand{\bfr}{\begin{frame}}
\newcommand{\efr}{\end{frame}}

\newcommand{\fti}[1]{\frametitle{#1}}

\newcommand{\pscb}[1]{\pscirclebox{#1}}

%\newcommand{\ch}{\choose}
\newcommand{\bib}{thebibliography}
\newcommand{\comp}{\models}
%\newcommand{\pf}{{\bf Proof.\hspace{5pt}}}
%\newcommand{\prf}{{\bf Proof\hspace{7pt}}}
%\newcommand{\qed}{\rule{1ex}{1ex}}
\newcommand{\hqed}{\hfill \qed}
%\newcommand{\hqedm}{\hfill \qed \medskip}
%\newcommand{\Qed}{\rule{1ex}{1ex} \medskip}
%\newcommand{\qqed}{\qquad\rule{1ex}{1ex}}
%\newcommand{\Qqed}{\qquad\rule{1ex}{1ex}\medskip}
%\newcommand{\eqed}[1]{$\textcolor{white}{\qed}\hfill{\dil#1}\hfill\qed$}
%\newcommand{\eqqed}[1]{$\rule{1ex}{0ex}\hfill{\dil#1}\hfill\qed$}
\newcommand{\mc}[3]{\multicolumn{#1}{#2}{#3}}
\newcommand{\Mc}[1]{\multicolumn{#1}{c}{}}
%\newcommand{\mc}[2]{\multicolumn{#1}{#2}}
\newcommand{\ul}{\underline}
\newcommand{\ol}{\overline}
\newcommand{\hor}{\mbox{--}}
\newcommand{\hs}[1]{\hspace{#1}}
\newcommand{\hso}[1]{\hspace{-1pt}}
\newcommand{\vs}[1]{\vspace{#1}}
\newcommand{\qmq}[1]{\quad\mbox{#1}\quad}
\newcommand{\ssk}{\smallskip}
\newcommand{\msk}{\mediumskip}
\newcommand{\bsk}{\bigskip}
\newcommand{\rp}[2]{\rule{#1pt}{#2pt}}
\newcommand{\str}[1]{\rule{#1pt}{0pt}}
%\newcommand{\st}[1]{\rule{#1}{0pt}}
\newcommand{\mbl}[1]{\makebox(0,0)[l]{#1}}
\newcommand{\mbr}[1]{\makebox(0,0)[r]{#1}}
\newcommand{\mbt}[1]{\makebox(0,0)[t]{#1}}
\newcommand{\mbb}[1]{\makebox(0,0)[b]{#1}}
\newcommand{\mbc}[1]{\makebox(0,0){#1}}
\newcommand\mar[1]{\marginpar{\raggedleft#1}}
\newcommand{\pmo}[2]{\put(#1){\makebox(0,0){#2}}}
%\newcommand{\new}[1]{\ensuremath{\blacktriangleright}#1\ensuremath{\blacktriangleleft}}
\newcommand{\emp}{\emptyset}
\newcommand{\indu}{\uparrow}
\newcommand{\rest}{\downarrow}
\newcommand{\sm}{\hspace{-2pt}\setminus\hspace{-2pt}}
\newcommand{\sbs}{\subset}
\newcommand{\sbe}{\subseteq}
\newcommand{\sps}{\supset}
\newcommand{\spe}{\supseteq}
\newcommand{\setm}{\setminus}
\newcommand{\asy}{\thicksim}
\newcommand{\dle}{<\hspace{-6pt}\cdot}
\newcommand{\dge}{\cdot\hspace{-6pt}>}
\newcommand{\lec}{\lessdot}
\newcommand{\grc}{\gtrdot}
\newcommand{\iso}{\cong}
%\newcommand{\con}{\equiv}
\newcommand{\Cong}{\equiv}
\newcommand{\conm}[1]{\stackrel{#1}{\equiv}}

\newcommand{\widevec}{\overrightarrow}
%makes a wide arrow over stuff if have \overrightarrow{stuff} 
\newcommand{\alh}{\widehat{\al}}
\newcommand{\Ch}{\widehat{C}}
\newcommand{\eh}{\widehat{e}}
\newcommand{\Gh}{\widehat{G}}
\newcommand{\ih}{\widehat{\imath}}
\newcommand{\jh}{\widehat{\jmath}}
\newcommand{\gh}{\widehat{g}}
\newcommand{\mh}{\widehat{0}}
\newcommand{\Mh}{\widehat{1}}
\newcommand{\Dh}{\widehat{D}}
\newcommand{\lah}{\widehat{\la}}
\newcommand{\zh}{\widehat{0}}
\newcommand{\oh}{\widehat{1}}
\newcommand{\Ph}{\widehat{P}}
\newcommand{\ph}{\widehat{p}}
\newcommand{\Pih}{\widehat{\Pi}}
\newcommand{\bbPh}{\widehat{\bbP}}
\newcommand{\Qh}{\widehat{Q}}
\newcommand{\Sh}{\widehat{S}}
\newcommand{\lh}{\widehat{l}}
\newcommand{\xh}{\widehat{x}}
\newcommand{\Yh}{\widehat{Y}}
\newcommand{\ptn}{\vdash}
\newcommand{\cpn}{\models}
\newcommand{\lt}{\lhd}
\newcommand{\gt}{\rhd}
\newcommand{\lte}{\unlhd}
\newcommand{\gte}{\unrhd}
\newcommand{\jn}{\vee}
\newcommand{\Jn}{\bigvee}
\newcommand{\mt}{\wedge}
\newcommand{\Mt}{\bigwedge}
\newcommand{\bdy}{\partial}
\newcommand{\sd}{\bigtriangleup}
\newcommand{\case}[4]{\left\{\overliner{ll}#1&\mbox{#2}\\#3&\mbox{#4}\earr\right.}
\newcommand{\fl}[1]{\lfloor #1 \rfloor}
\newcommand{\ce}[1]{\lceil #1 \rceil}
\newcommand{\flf}[2]{\left\lfloor\frac{#1}{#2}\right\rfloor}
\newcommand{\cef}[2]{\left\lceil\frac{#1}{#2}\right\rceil}
\newcommand{\gau}[2]{\left[ \overliner{c} #1 \\ #2 \earr \right]}
\newcommand{\gauss}[2]{\genfrac{[}{]}{0pt}{}{#1}{#2}} 
\newcommand{\setc}[2]{\{left{#1}\ :\ #2\}}
\newcommand{\setl}[2]{\{left{#1}\ |\ #2\}}
\newcommand{\qst}{$q$-Stirling numbers}
\newcommand{\qbi}{$q$-binomial coefficients}
\newcommand{\del}{\nabla}
\newcommand{\pde}[2]{\frac{\partial#1}{\partial#2}}
\newcommand{\spd}[2]{\frac{\partial^2\hspace{-2pt}#1}{\partial#2^2}}
\newcommand{\ode}[2]{\ds\frac{d#1}{d#2}}
\def\<{\langle}
\def\>{\rangle}
\newcommand{\fall}[2]{\langle{#1}\rangle_{#2}}
\newcommand{\nor}[1]{\parallel{#1}\parallel}
\newcommand{\ipr}[2]{\langle{#1},{#2}\rangle}
\newcommand{\spn}[1]{\langle{#1}\rangle}
%\newcommand{\lb}{\left\{}
%\newcommand{\rb}{\right\}}
\newcommand{\ree}[1]{(\ref{#1})}
\newcommand{\rpl}{$\longleftarrow$}

\newcommand{\ra}{\rightarrow}
%\newcommand{\Ra}{\Rightarrow}
\newcommand{\lta}{\leftarrow}
\newcommand{\rla}{\leftrightarrow}
\newcommand{\lra}{\longrightarrow}
\newcommand{\lla}{\longleftarrow}
\newcommand{\Lra}{\Longrightarrow}
\newcommand{\llra}{\longleftrightarrow}
\newcommand{\Llra}{\Longleftrightarrow}
\newcommand{\ve}[1]{\stackrel{\ra}{#1}}


\newcommand{\al}{\alpha}
\newcommand{\be}{\beta}
\newcommand{\ga}{\gamma}
\newcommand{\de}{\delta}
\newcommand{\ep}{\epsilon}
\newcommand{\vep}{\varepsilon}
\newcommand{\io}{\iota}
%%% A reviewer suggested we avoid kappa, since it has a rowmotion-related meaning elsewhere
%%% in lattice theory; Changing all \ka and \kappa to \orevbi, and redefining that to \xi 
\def\orevbi{\beta } 
\newcommand{\ka}{\kappa}
\newcommand{\la}{\lambda}
\newcommand{\mut}{\widetilde{\mu}}
\newcommand{\om}{\omega}
\newcommand{\Phit}{\widetilde{\Phi}}
\newcommand{\Psit}{\widetilde{\Psi}}
\newcommand{\rhot}{\widetilde{\rho}}
\newcommand{\si}{\sigma}
\renewcommand{\th}{\theta}
%       NOTE THAT \th HAS BEEN RENEWCOMMANDed
\newcommand{\ze}{\zeta}
\newcommand{\Al}{\Alpha}
\newcommand{\Ga}{\Gamma}
\newcommand{\De}{\Delta}
\newcommand{\Ep}{\epsilon}
\newcommand{\La}{\Lambda}
\newcommand{\Om}{\Omega}
\newcommand{\Si}{\Sigma}
\newcommand{\Th}{\Theta}
\newcommand{\Ze}{\Zeta}


\newcommand{\0}{{\bf 0}}
\newcommand{\1}{{\bf 1}}
\newcommand{\2}{{\bf 2}}
\newcommand{\3}{{\bf 3}}
\newcommand{\4}{{\bf 4}}
\newcommand{\5}{{\bf 5}}
\newcommand{\6}{{\bf 6}}
\newcommand{\7}{{\bf 7}}
\newcommand{\8}{{\bf 8}}
\newcommand{\9}{{\bf 9}}
\newcommand{\ba}{{\bf a}}
\newcommand{\bb}{{\bf b}}
\newcommand{\bc}{{\bf c}}
\newcommand{\bd}{{\bf d}}
\newcommand{\bfe}{{\bf e}}
\newcommand{\bff}{{\bf f}}
\newcommand{\bg}{{\bf g}}
\newcommand{\bh}{{\bf h}}
\newcommand{\bi}{{\bf i}}
\newcommand{\bj}{{\bf j}}
\newcommand{\bk}{{\bf k}}
\newcommand{\bl}{{\bf l}}
\newcommand{\bm}{{\bf m}}
\newcommand{\bn}{{\bf n}}
\newcommand{\bp}{{\bf p}}
\newcommand{\bq}{{\bf q}}
\newcommand{\br}{{\bf r}}
\newcommand{\bs}{{\bf s}}
\newcommand{\bt}{{\bf t}}
\newcommand{\bu}{{\bf u}}
\newcommand{\bv}{{\bf v}}
\newcommand{\bw}{{\bf w}}
\newcommand{\bx}{{\bf x}}
\newcommand{\by}{{\bf y}}
\newcommand{\bz}{{\bf z}}
\newcommand{\bA}{{\bf A}}
\newcommand{\bB}{{\bf B}}
\newcommand{\bC}{{\bf C}}
\newcommand{\bE}{{\bf E}}
\newcommand{\bF}{{\bf F}}
\newcommand{\bG}{{\bf G}}
\newcommand{\bH}{{\bf H}}
\newcommand{\bN}{{\bf N}}
\newcommand{\bP}{{\bf P}}
\newcommand{\bQ}{{\bf Q}}
\newcommand{\bR}{{\bf R}}
\newcommand{\bS}{{\bf S}}
\newcommand{\bT}{{\bf T}}
\newcommand{\bV}{{\bf V}}
\newcommand{\bX}{{\bf X}}
\newcommand{\bXh}{\widehat{{\bf X}}}
\newcommand{\bZ}{{\bf Z}}
\newcommand{\bep}{{\bf \ep}}
\newcommand{\bcdot}{{\bf \cdot}}
\newcommand{\bmpl}{
\hspace{2pt}
\begin{pspicture}(.3,.3)
\psline[linewidth=1.5pt,arrows=cc-cc](.15,0)(.15,.3)
\psline[linewidth=1.5pt,arrows=cc-cc](0,.15)(.3,.15)
\end{pspicture}
\hspace{2pt}
}
\newcommand{\bmmi}{
\hspace{2pt}
\begin{pspicture}(.3,.3)
\psline[linewidth=1.5pt,arrows=cc-cc](0,.15)(.3,.15)
\end{pspicture}
\hspace{2pt}
}
\newcommand{\bma}{\mbox{\boldmath $a$}}
\newcommand{\bmb}{\mbox{\boldmath $b$}}
\newcommand{\bmk}{\mbox{\boldmath $k$}}
\newcommand{\bml}{\mbox{\boldmath $l$}}
\newcommand{\bmn}{\mbox{\boldmath $n$}}
\newcommand{\bbC}{{\mathbb C}}
\newcommand{\bbF}{{\mathbb F}}
\newcommand{\bbM}{{\mathbb M}}
\newcommand{\bbN}{{\mathbb N}}
\newcommand{\bbP}{{\mathbb P}}
\newcommand{\bbQ}{{\mathbb Q}}
\newcommand{\bbR}{{\mathbb R}}
\newcommand{\bbS}{{\mathbb S}}
\newcommand{\bbZ}{{\mathbb Z}}
\newcommand{\cA}{{\cal A}}
\newcommand{\cAB}{{\cal AB}}
\newcommand{\cAD}{{\cal AD}}
\newcommand{\cB}{{\cal B}}
\newcommand{\cBC}{{\cal BC}}
\newcommand{\cC}{{\cal C}}
\newcommand{\cD}{{\cal D}}
\newcommand{\cDB}{{\cal DB}}
\newcommand{\cE}{{\cal E}}
\newcommand{\cF}{{\cal F}}
\newcommand{\cG}{{\cal G}}
\newcommand{\cH}{{\cal H}}
\newcommand{\cI}{{\cal I}}
\newcommand{\cIF}{{\cal IF}}
\newcommand{\cJ}{{\cal J}}
\newcommand{\cK}{{\cal K}}
\newcommand{\cL}{{\cal L}}
\newcommand{\cm}{{\cal m}}
\newcommand{\cM}{{\cal M}}
\newcommand{\cN}{{\cal N}}
\newcommand{\cNBC}{{\cal NBC}}
\newcommand{\cO}{{\cal O}}
\newcommand{\cP}{{\cal P}}
\newcommand{\cPD}{{\cal PD}}
\newcommand{\cR}{{\cal R}}
\newcommand{\cRF}{{\cal RF}}
\newcommand{\cS}{{\cal S}}
\newcommand{\cT}{{\cal T}}
\newcommand{\cTb}{\ol{\cal T}}
\newcommand{\cV}{{\cal V}}
\newcommand{\cW}{{\cal W}}
\newcommand{\cX}{{\cal X}}
\newcommand{\cY}{{\cal Y}}
\newcommand{\cZ}{{\cal Z}}

\newcommand{\df}{\dot{f}}
\newcommand{\dF}{\dot{F}}
\newcommand{\fg}{{\mathfrak g}}
\newcommand{\fG}{{\mathfrak G}}
\newcommand{\fgt}{\widetilde{\mathfrak g}}
\newcommand{\fgl}{{\mathfrak gl}}
\newcommand{\fL}{{\mathfrak L}}
\newcommand{\fS}{{\mathfrak S}}
\newcommand{\fsl}{{\mathfrak sl}}
\newcommand{\at}{\widetilde{a}}
\newcommand{\ct}{\widetilde{c}}
\newcommand{\dt}{\widetilde{d}}
\newcommand{\pt}{\widetilde{p}}
\newcommand{\Bt}{\widetilde{B}}
\newcommand{\Gt}{\widetilde{G}}
\newcommand{\Ht}{\widetilde{H}}
\newcommand{\Kt}{\widetilde{K}}
\newcommand{\Nt}{\widetilde{N}}
\newcommand{\St}{\widetilde{S}}
\newcommand{\Xt}{\widetilde{X}}
\newcommand{\alt}{\widetilde{\alpha}}
\newcommand{\bet}{\widetilde{\beta}}
\newcommand{\rht}{\widetilde{\rho}}
\newcommand{\Pit}{\widetilde{\Pi}}
\newcommand{\tal}{\widetilde{\alpha}}
\newcommand{\tbe}{\widetilde{\beta}}
\newcommand{\tPi}{\widetilde{\Pi}}
\newcommand{\ab}{\ol{a}}
\newcommand{\Ab}{\ol{A}}
\newcommand{\Bb}{\ol{B}}
\newcommand{\cb}{\ol{c}}
\newcommand{\Cb}{{\ol{C}}}
\newcommand{\db}{\ol{d}}
\newcommand{\Db}{\ol{D}}
\newcommand{\eb}{\ol{e}}
\newcommand{\Eb}{\ol{E}}
\newcommand{\fb}{\ol{f}}
\newcommand{\Fb}{\ol{F}}
\newcommand{\Gb}{\ol{G}}
\newcommand{\Hb}{\ol{H}}
\newcommand{\ib}{\ol{\imath}}
\newcommand{\Ib}{\ol{I}}
\newcommand{\jb}{\ol{\jmath}}
\newcommand{\Jb}{\ol{J}}
\newcommand{\kb}{\ol{k}}
\newcommand{\Kb}{\ol{K}}
\newcommand{\nb}{\ol{n}}
\newcommand{\pb}{\ol{p}}
\newcommand{\Pb}{\ol{P}}
\newcommand{\Phib}{\ol{\Phi}}
\newcommand{\Qb}{\ol{Q}}
\newcommand{\rb}{\ol{r}}
\newcommand{\Rb}{\ol{R}}
\newcommand{\Sb}{\ol{S}}
\newcommand{\tb}{\ol{t}}
\newcommand{\Tb}{\ol{T}}
\newcommand{\Ub}{\ol{U}}
\newcommand{\Wb}{\ol{W}}
\newcommand{\xb}{\ol{x}}
\newcommand{\Xb}{\ol{X}}
\newcommand{\yb}{\ol{y}}
\newcommand{\Yb}{\ol{Y}}
\newcommand{\zb}{\ol{z}}
\newcommand{\Zb}{\ol{Z}}
\newcommand{\pib}{\ol{\pi}}
\newcommand{\sib}{\ol{\si}}
\newcommand{\degb}{\ol{\deg}}
\newcommand{\vj}{\vec{\jmath}}
\newcommand{\vr}{\vec{r}}
\newcommand{\vv}{\vec{v}}
\newcommand{\vw}{\vec{w}}
\newcommand{\ttab}{\{t\}}
\newcommand{\stab}{\{s\}}


\DeclareMathOperator{\Age}{Age}
\DeclareMathOperator{\Aut}{Aut}
\DeclareMathOperator{\Av}{Av}
\DeclareMathOperator{\abl}{al}
\DeclareMathOperator{\asc}{asc}
\DeclareMathOperator{\BC}{BC}
\DeclareMathOperator{\capa}{cap}
\DeclareMathOperator{\codim}{codim}
\DeclareMathOperator{\ch}{ch}
%Commented out \ch for choose
\DeclareMathOperator{\col}{col}
\DeclareMathOperator{\ctt}{ct}
\DeclareMathOperator{\Der}{Der}
\DeclareMathOperator{\des}{des}
\DeclareMathOperator{\Des}{Des}
\DeclareMathOperator{\Det}{Det}
\DeclareMathOperator{\diag}{diag}
\DeclareMathOperator{\Diag}{Diag}
\DeclareMathOperator{\diam}{diam}
\DeclareMathOperator{\End}{End}
\DeclareMathOperator{\ev}{ev}
\DeclareMathOperator{\eval}{eval}
\DeclareMathOperator{\Eval}{Eval}
\DeclareMathOperator{\FPF}{FPF}
\DeclareMathOperator{\Frac}{Frac}
\DeclareMathOperator{\gen}{gen}
\DeclareMathOperator{\GL}{GL}
\DeclareMathOperator{\Harm}{Harm}
\DeclareMathOperator{\Hom}{Hom}
\DeclareMathOperator{\id}{id}
\DeclareMathOperator{\im}{im}
\DeclareMathOperator{\imaj}{imaj}
\DeclareMathOperator{\inc}{inc}
\DeclareMathOperator{\Ind}{Ind}
\DeclareMathOperator{\inter}{int}
\DeclareMathOperator{\Inter}{Int}
\DeclareMathOperator{\inv}{inv}
\DeclareMathOperator{\Inv}{Inv}
\DeclareMathOperator{\lcm}{lcm}
\DeclareMathOperator{\LHS}{LHS}
\DeclareMathOperator{\Mat}{Mat}
\DeclareMathOperator{\maj}{maj}
\DeclareMathOperator{\mdeg}{mdeg}
\DeclareMathOperator{\Mod}{mod}
\DeclareMathOperator{\NBB}{NBB}
\DeclareMathOperator{\nin}{nin}
\DeclareMathOperator{\Nin}{Nin}
\DeclareMathOperator{\nul}{nul}
\DeclareMathOperator{\nullity}{nullity}
\DeclareMathOperator{\NSym}{{\bf NSym}}
\DeclareMathOperator{\od}{od}
\DeclareMathOperator{\ord}{ord}
\DeclareMathOperator{\per}{per}
\DeclareMathOperator{\Pete}{Pete}
\DeclareMathOperator{\pfa}{pf}
\DeclareMathOperator{\Pm}{pm}
\DeclareMathOperator{\PM}{PM}
\DeclareMathOperator{\Pol}{Pol}
\DeclareMathOperator{\rad}{rad}
\DeclareMathOperator{\RHS}{RHS}
\DeclareMathOperator{\rk}{rk}
\DeclareMathOperator{\rank}{rank}
\DeclareMathOperator{\reg}{reg}
\DeclareMathOperator{\rlm}{rlm}
\DeclareMathOperator{\row}{row}
\DeclareMathOperator{\sdiam}{sdiam}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\sign}{sign}
\DeclareMathOperator{\sh}{sh}
\DeclareMathOperator{\slack}{slack}
\DeclareMathOperator{\spa}{span}
\DeclareMathOperator{\st}{st}
\DeclareMathOperator{\summ}{sum}
\DeclareMathOperator{\Supp}{Supp}
\DeclareMathOperator{\sus}{susp}
\DeclareMathOperator{\Sym}{{\bf Sym}}
\DeclareMathOperator{\SYT}{SYT}
\DeclareMathOperator{\tang}{tang}
\DeclareMathOperator{\tr}{tr}
\DeclareMathOperator{\tri}{tri}
\DeclareMathOperator{\wed}{wedge}
\DeclareMathOperator{\wt}{wt}







\newcommand{\new}[1]{\vs{5pt}\ensuremath{\blacktriangleright}#1\ensuremath{\blacktriangleleft}\vs{5pt}}
\newcommand{\bul}{\bullet}
\newcommand{\dil}{\displaystyle}
\newcommand{\dsum}{\dil\sum}
\newcommand{\dint}{\dil\int}
%\newcommand{\dfrac}{\dil\frac}
\newcommand{\scl}{\scriptstyle}
\newcommand{\ssl}{\scriptscriptstyle}
\newcommand{\foz}{\footnotesize}
\newcommand{\scz}{\scriptsize}
\newcommand{\cho}{\choose}

\newcommand{\Schu}{Sch\"utzenberger}
\newcommand{\aim}{Adv.\ in Math.\/}
\newcommand{\bams}{Bull.\ Amer.\ Math.\ Soc.\/}
\newcommand{\cjm}{Canad.\ J. Math.\/}
\newcommand{\dm}{Discrete Math.\/}
\newcommand{\dmj}{Duke Math.\ J.\/}
\newcommand{\ejc}{European J. Combin.\/}
\newcommand{\jaa}{J. Algebra\/}
\newcommand{\jac}{J. Algebraic Combin.\/}
\newcommand{\jas}{J. Algorithms\/}
\newcommand{\jams}{J. Amer.\ Math.\ Soc.\/}
\newcommand{\jct}{J. Combin.\ Theory\/}
\newcommand{\jcta}{J. Combin.\ Theory Ser. A\/}
\newcommand{\jctb}{J. Combin.\ Theory Ser. B\/}
\newcommand{\jgt}{J. Graph Theory\/}
\newcommand{\jram}{J. Reine Angew.\ Math.\/}
\newcommand{\pjm}{Pacific J. Math.\/}
\newcommand{\pams}{Proc.\ Amer.\ Math.\ Soc.\/}
\newcommand{\plms}{Proc.\ London Math.\ Soc.\/}
\newcommand{\tams}{Trans.\ Amer.\ Math.\ Soc.\/}
\newcommand{\pja}{Proc.\ Japan Acad.\ Ser.\ A  Math\/}
\newcommand{\sv}{Springer-Verlag Lecture Notes in Math.\/}
\newcommand{\crgs}{Combinatoire et Repr\'{e}sentation du 
    Groupe Sym\'{e}trique, Strasbourg 1976, D. Foata ed.}
\newcommand{\caup}{Cambridge University Press}
\newcommand{\oup}{Oxford University Press} 
\newcommand{\pr}{preprint}
\newcommand{\ip}{in preparation}
\newcommand{\ds}{\displaystyle}


\begin{document}
\begin{thebibliography}{99}

% \bibitem{CF:oar}
% P.~J. Cameron and D.~G. Fon-Der-Flaass.
% \newblock Orbits of antichains revisited.
% \newblock {\em European J. Combin.}, 16(6):545--554, 1995.

% \bibitem{cla:epp}
% Andrew Claussen.
% \newblock {\em Expansion posets for polygon cluster algebras}.
% \newblock PhD thesis, Michigan State University, 2020.

% \bibitem{DPS:rop}
% Kevin Dilks, Oliver Pechenik, and Jessica Striker.
% \newblock Resonance in orbits of plane partitions and increasing tableaux.
% \newblock {\em J. Combin. Theory Ser. A}, 148:244--274, 2017.

% \bibitem{EE:cce}
% Henrik Eriksson and Kimmo Eriksson.
% \newblock Conjugacy of {C}oxeter elements.
% \newblock {\em Electron. J. Combin.}, 16(2, Special volume in honor of Anders
%   Bj\"{o}rner):Research Paper 4, 7, 2009.

% \bibitem{EP:pbt}
% David Einstein and James Propp.
% \newblock Piecewise-linear and birational toggling.
% \newblock In {\em 26th {I}nternational {C}onference on {F}ormal {P}ower
%   {S}eries and {A}lgebraic {C}ombinatorics ({FPSAC} 2014)}, Discrete Math.
%   Theor. Comput. Sci. Proc., AT, pages 513--524. Assoc. Discrete Math. Theor.
%   Comput. Sci., Nancy, 2014.

% \bibitem{EP:cpbh}
% David Einstein and James Propp.
% \newblock Combinatorial, piecewise-linear, and birational homomesy for products
%   of two chains.
% \newblock {\em Algebr. Comb.}, 4(2):201--224, 2021.

% \bibitem{ES:prs}
% Sergi Elizalde and Bruce~E. Sagan.
% \newblock Partial rank symmetry and fences.
% \newblock In preparation.

% \bibitem{gan:loi}
% Emden~R. Gansner.
% \newblock On the lattice of order ideals of an up-down poset.
% \newblock {\em Discrete Math.}, 39(2):113--122, 1982.

% \bibitem{GP:rs}
% Pavel Galashin and Pavlo Pylyavskyy.
% \newblock {$R$}-systems.
% \newblock {\em Selecta Math. (N.S.)}, 25(2):Paper No. 22, 63, 2019.

% \bibitem{GR:ipb2}
% Darij Grinberg and Tom Roby.
% \newblock Iterative properties of birational rowmotion {II}: rectangles and
%   triangles.
% \newblock {\em Electron. J. Combin.}, 22(3):Paper 3.40, 49, 2015.

% \bibitem{GR:ipb1}
% Darij Grinberg and Tom Roby.
% \newblock Iterative properties of birational rowmotion {I}: generalities and
%   skeletal posets.
% \newblock {\em Electron. J. Combin.}, 23(1):Paper 1.33, 40, 2016.

% \bibitem{jos:atr}
% Michael Joseph.
% \newblock Antichain toggling and rowmotion.
% \newblock {\em Electron. J. Combin.}, 26(1):Paper No. 1.29, 43, 2019.

% \bibitem{JR:tispg}
% Michael Joseph and Tom Roby.
% \newblock Toggling independent sets of a path graph.
% \newblock {\em Electron. J. Combin.}, 25(1):Paper No. 1.18, 31, 2018.

% \bibitem{MGO:qdr}
% Sophie Morier-Genoud and Valentin Ovsienko.
% \newblock {$q$}-deformed rationals and {$q$}-continued fractions.
% \newblock {\em Forum Math. Sigma}, 8:Paper No. e13, 55, 2020.

% \bibitem{MR:pub}
% Gregg Musiker and Tom Roby.
% \newblock Paths to understanding birational rowmotion on products of two
%   chains.
% \newblock {\em Algebr. Comb.}, 2(2):275--304, 2019.

% \bibitem{MSS:ruc}
% Thomas McConville, Bruce~E. Sagan, and Clifford Smyth.
% \newblock On a rank-unimodality conjecture of {M}orier-{G}enoud and {O}vsienko.
% \newblock {\em Discrete Math.}, 344(8):112483, 13, 2021.

% \bibitem{MSW:pca}
% Gregg Musiker, Ralf Schiffler, and Lauren Williams.
% \newblock Positivity for cluster algebras from surfaces.
% \newblock {\em Adv. Math.}, 227(6):2241--2308, 2011.

% \bibitem{MZ:rpl}
% Emanuele Munarini and Norma Zagaglia~Salvi.
% \newblock On the rank polynomial of the lattice of order ideals of fences and
%   crowns.
% \newblock {\em Discrete Math.}, 259(1-3):163--177, 2002.

% \bibitem{pro:cfp}
% James Propp.
% \newblock The combinatorics of frieze patterns and {M}arkoff numbers.
% \newblock {\em Integers}, 20:Paper No. A12, 38, 2020.

% \bibitem{rob:dac}
% Tom Roby.
% \newblock Dynamical algebraic combinatorics and the homomesy phenomenon.
% \newblock In {\em Recent trends in combinatorics}, volume 159 of {\em IMA Vol.
%   Math. Appl.}, pages 619--652. Springer, [Cham], 2016.

% \bibitem{sch:cef}
% Ralf Schiffler.
% \newblock A cluster expansion formula ({$A_n$} case).
% \newblock {\em Electron. J. Combin.}, 15(1):Research paper 64, 9, 2008.

% \bibitem{sch:caa}
% Ralf Schiffler.
% \newblock On cluster algebras arising from unpunctured surfaces. {II}.
% \newblock {\em Adv. Math.}, 223(6):1885--1923, 2010.

% \bibitem{ST:caa}
% Ralf Schiffler and Hugh Thomas.
% \newblock On cluster algebras arising from unpunctured surfaces.
% \newblock {\em Int. Math. Res. Not. IMRN}, (17):3160--3189, 2009.

% \bibitem{str:dac}
% Jessica Striker.
% \newblock Dynamical algebraic combinatorics: promotion, rowmotion, and
%   resonance.
% \newblock {\em Notices Amer. Math. Soc.}, 64(6):543--549, 2017.

% \bibitem{str:rgt}
% Jessica Striker.
% \newblock Rowmotion and generalized toggle groups.
% \newblock {\em Discrete Math. Theor. Comput. Sci.}, 20(1):Paper No. 17, 26,
%   2018.

% \bibitem{SW:pr}
% Jessica Striker and Nathan Williams.
% \newblock Promotion and rowmotion.
% \newblock {\em European J. Combin.}, 33(8):1919--1942, 2012.

% \bibitem{TW:rsm}
% H.~Thomas and N.~Williams.
% \newblock Rowmotion in slow motion.
% \newblock {\em Proc. Lond. Math. Soc. (3)}, 119(5):1149--1178, 2019.

% \bibitem{vor:hpt}
% Corey Vorland.
% \newblock Homomesy in products of three chains and multidimensional
%   recombination.
% \newblock {\em Electron. J. Combin.}, 26(4):Paper No. 4.30, 26, 2019.

% \bibitem{yur:cef}
% Toshiya Yurikusa.
% \newblock Cluster expansion formulas in type {A}.
% \newblock {\em Algebr. Represent. Theory}, 22(1):1--19, 2019.

% \bibitem{yur:cce}
% Toshiya Yurikusa.
% \newblock Combinatorial cluster expansion formulas from triangulated surfaces.
% \newblock {\em Electron. J. Combin.}, 26(2):Paper No. 2.33, 39, 2019.

% \end{thebibliography}


\maketitle
%% note that you DO NOT have to put your abstract here -- it is generated by \maketitle and the \abstract and \resume commands above


\section{Introduction}

%The purpose of this work is to
We initiate the study of the dynamical algebraic combinatorics of fence posets.  A {\em fence} is a poset with elements $F=\{x_1,x_2,\ldots,x_n\}$, partial order $\lte$, and covers
\vspace{-2pt}
\[
x_1\lt x_2 \lt \cdots \lt x_a \gt x_{a+1} \gt \cdots \gt x_b \lt x_{b+1} \lt \cdots,
\vspace{-2pt}
\]
where $a,b,\ldots$ are positive integers.  The maximal chains of $F$ are called {\em segments}.   A fence with seven elements and three segments is shown in Figure~\ref{F(3,3,2)}.  
Throughout we will often use $n$ for the cardinality of $F$ which we denote by $\#F$. 
We also let $[n]=\{1,2,\ldots,n\}$.

There are a number of different conventions for indicating the size of the segments of a fence $F$ in the literature depending on the application being considered.  Our results will be simplest if described in terms of unshared elements. 
Call $x\in F$ {\em shared} if it is the intersection of two segments; otherwise $x$ is {\em unshared}.  In Figure~\ref{F(3,3,2)}, the elements $x_3$ and $x_6$ are shared and all other elements unshared.  Let $\al=(\al_1,\al_2,\ldots,\al_s)$ be a composition (a sequence of positive integers) with $\al_1,\al_s\ge2$.  The corresponding fence is 
$ 
F=\uF(\al)=\uF(\al_1,\al_2,\ldots,\al_s),
$
where
$
\al_i = 1+(\text{number of unshared elements on the $i$th segment})
$
for $i\in[s]$.  The fence in Figure~\ref{F(3,3,2)} is $F=\uF(3,3,2)$.
For any fence, 
$
\#F=\al_1+\al_2+\cdots+\al_s-1.
$


\begin{wrapfigure}[7]{r}{165pt}\centering
%\bfi
\begin{tikzpicture}[scale = 0.5]
\fill(0,1) circle(.1);
\fill(1,2) circle(.1);
\fill(2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,0) circle(.1);
\fill(6,1) circle(.1);
\draw (0,1)--(2,3)--(5,0)--(6,1);
\draw (0,.5) node{$x_1$};
\draw (1,1.5) node{$x_2$};
\draw (2,2.5) node{$x_3$};
\draw (3,1.5) node{$x_4$};
\draw (4,.5) node{$x_5$};
\draw (5,-.5) node{$x_6$};
\draw (6,.5) node{$x_7$};
\end{tikzpicture}
\capt{The fence $F=\uF(3,3,2)$ \label{F(3,3,2)}}
%\efi
\end{wrapfigure}


Fences have been of recent interest because of their connections with cluster algebras, $q$-analogues, and unimodality.  
Let $P$ be a poset with partial order $\lte$. Recall that $I\sbe P$ is a {\em (lower order) ideal} of $P$ if $x\in I$ and $y\lte x$ imply $y\in I$. {\em Upper order ideals}, $U$, are defined by reversing the inequality. 
When merely writing ``ideal,'' we always mean a lower order ideal.
Let $\cI(P)$ and $\cU(P)$ be the set of (lower order) ideals and upper order ideals of $P$, respectively. 
The set of ideals of a finite poset forms a distributive lattice under inclusion. 
%If $F$ is a fence then we denote this lattice by $\cI(F)$.
If $P$ is a fence then this lattice can be used to calculate the mutations
 in a cluster algebra derived from  a surface with marked points on the boundary~\cite{cla:epp,MSW:pca,pro:cfp,sch:caa,ST:caa,yur:cce}. 
%%% OMITTED yur:cef for space, which is sited in yur:cce

Let $q$ be a variable and $r(F;q)$ be the rank generating function for $\cI(F)$.
Mourier-Genoud and Ovsienko~\cite{MGO:qdr} were able to define $q$-analogues of rational
numbers which are certain rational functions of $q$.  The numerators and
denominators of these fractions are exactly the polynomials $r(F;q)$, which they conjectured were unimodal.   
%%%In addition, they made the following intriguing conjecture.  
Progress on this question can be found
in~\cite{cla:epp,ES:prs,gan:loi,MSS:ruc,MZ:rpl}. Quite recently a proof was found by Ezgi
Kantarc\i~O\u guz and Mohan Ravichandran~\cite[]{OR:rpfpu}.   
% \bcon[\cite{MGO:qdr}] The polynomial
% $r(F;q)$ is unimodal.  \econ

Our  focus is going to be on algebraic dynamics of fences and, in particular, on rowmotion.
A subset $A$ of a finite poset $(P,\lte)$ is an {\em antichain} if no two elements of $A$ are comparable.
Let $\cA(P)$ be the set of antichains of $P$.  
%Let $\cU(P)$ and $\cA(P)$ be the set of upper order ideals and antichains of $P$, respectively.  
Ideals of both types and antichains are related by the maps
$\De \colon \cI(P)\ra\cA(P)$ where
$
\De(I) = \{x\in I \mid \text{$x$ is a maximal element of $I$}\},
$
and $\na\colon \cU(P)\ra\cA(P)$ where
$
\na(U) = \{x\in U \mid \text{$x$ is a minimal element of $U$}\}.
$
We also let $c\colon \cP\ra\cP$ be the complement operator $c(S)=P-S$.
{\em Rowmotion on antichains} is the group action on $\cA(P)$ generated by the map
$
\rho = \na \circ c \circ \De^{-1}
$
where we always compose functions from right to left.  We will also consider {\em rowmotion on ideals}, which is generated by
$
\rhoh =  \De^{-1} \circ \na \circ c.
$
There is clearly a bijection between the orbits of $\rho$ and those of $\rhoh$.
We will call the number of elements (that is, the number of antichains or, equivalently, the number of ideals) in an orbit either its {\em size} or its {\em length}.
Rowmotion and its generalizations have been investigated by many authors, for example~\cite{CF:oar,PR:hp2c,SW:pr,TW:rsm}.  See, in particular, the survey articles of Roby~\cite{rob:dac} and Striker~\cite{str:dac} and the references therein.
We will let 
$
\cI(\al)=\cI(\uF(\al))
$ 
and similarly for $\cU$ and $\cA$.

In addition to describing the orbits of rowmotion on fences, we will also consider properties of various statistics.
If $S$ is a finite set, then a 
{\em statistic} on $S$ is a map
$\st\colon S\ra\bbN$ where $\bbN$ is the set of  nonnegative integers.
If $G$ is a group acting on $S$, then statistic $\st$ is 
{\em $d$-mesic}  if there is a constant $d$ such that every orbit $\cO$ of the action has average
$
\frac{\st\cO}{\#\cO}=d,
$
where $\st\cO=\sum_{x\in\cO} \st x$ and $\#\cO$ is the size of the orbit.
We say that $\st$ is {\em homomesic} if it is $d$-mesic for some $d$.  Homomesy is a
well-studied property of rowmotion; see~\cite{PR:hp2c} for examples.  Rowmotion on fences displays a new and interesting
phenomenon.  We say that $\st$ is {\em homometric} (FKA \emph{orbomesic}) if 
$\st\cO$ is constant over all orbits of the same cardinality.  Equivalently, for any two orbits $\cO_1$ and $\cO_2$ we have
$
\#\cO_1 = \#\cO_2 \implies 
\st\cO_1 =\st\cO_2.
$
Note that homomesy implies homometry, but not conversely.
In the sequel we will see many homometries which are not homomesies.

One can break down rowmotion into smaller steps called toggles.  Let  $S$ be a finite set and let $\cF(S)$ be a family of subsets of $S$.  If $x\in S$ then the corresponding {\em toggle map} is $\tau_x\colon \cF(S)\ra\cF(S)$ defined on $T\in\cF(S)$ by
$
\tau_x(T) =
\begin{cases}
T\oplus\{x\} &\text{if $T\oplus\{x\}\in\cF(S)$,}\\
T           &\text{else,}
\end{cases}
$
where $\oplus$ is symmetric difference.  Clearly $\tau_x$ is an involution. 
The corresponding {\em toggle group} is the group $\cT$ generated by the toggles $\tau_x$ for $x\in S$.  
A {\em Coxeter element} in $\cT$ is a product
$\tau_{x_1} \tau_{x_2}\cdots\tau_{x_n}$ where 
$x_1,x_2,\ldots,x_n$ is some permutation of $S$.
If $P$ is a finite poset and $x\in P$, then we denote by $\tau_x$ and $\tauh_x$ the corresponding toggles on $\cA(P)$ and $\cI(P)$, respectively.
We will let $\cT_P$ and $\cTh_P$ denote the toggle groups generated by the toggles
$\tau_x$ and $\tauh_x$, respectively.
The relation of toggles, and in particular the corresponding Coxeter elements, to rowmotion is given in the following fundamental result of Cameron and Fon-Der-Flaass.
\begin{thm}[\cite{CF:oar}] \label{thm:rowcox}
Let $P$ be a finite poset and let $x_1,x_2,\ldots,x_n$ be any linear extension of $P$.  Then
$\rhoh = \tauh_{x_1}\tauh_{x_2}\ldots\tauh_{x_n}.$

\end{thm}


The rest of this paper is structured as follows.
In the next section we introduce our principal tool in this work, which is a representation of rowmotion orbits of antichains of fences in terms of certain tilings.
Section~\ref{gf} is devoted to applying this model to prove various homomesy results for fences with any number of segments.  We also define, for any self-dual poset, a new orbit structure which is coarser than that of rowmotion, and prove a homomesy on ideals in this setting.
Section~\ref{ffs} is devoted to examining orbits and homometries for certain fences having at
most five segments.  In Section~\ref{tog} we prove a general result for Coxeter elements and
associated homomesies which applies to $\cTh_F$ for any fence $F$.  We end with a section
containing conjectures and future directions. 

%%%ADDED for FPSAC
This extended abstract is a summary of result from our preprint~\cite{EPRS:rf}, where full
details (including the omitted proofs) and a few further results can be found.  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Tilings}
\label{til}

It turns out that the orbits of $\cA(\al)$ can be nicely visualized in terms of tilings.  This will be our principal tool in proving homo- and homometry results.

 


Consider a rectangle $R_s$ subdivided into unit squares with rows  numbered $1,2,\ldots,s$ from top to bottom, and infinitely many columns. 
The rows will correspond to the segments of a corresponding fence and the columns to antichains in an orbit.  For example, $R_4$ is
$$
\begin{tikzpicture}[scale=.4]
\draw[very thin] (0,0) grid (10,4);
\draw (-1,2) node{$\cdots$};
\draw (11,2) node{$\cdots$};
\draw(-3,3.5) node{$1$};
\draw(-3,2.5) node{$2$};
\draw(-3,1.5) node{$3$};
\draw(-3,.5) node{$4$};
\end{tikzpicture}
$$

 In the following definition, an {\em $a\times b$ tile} is a tile which covers $a$ rows and $b$ columns in $R_s$.

\begin{definition}
\label{TilDef}
If $\al=(\al_1,\al_2,\ldots,\al_s)$, then an {\em $\al$-tiling} is a tiling of $R_s$  using yellow $1\times 1$ tiles, red $2\times 1$ tiles, and black $1\times(\al_i-1)$ tiles in row $i$, for $1\le i\le s$, such that the following hold for all rows.
\begin{enumerate}
    \item[(a)] If $\al_i\ge2$ and the red tiles are ignored, then the black and yellow  tiles alternate in  row $i$.
    \item[(b)] There is a red tile in a column covering rows $i$ and $i+1$ if and only if either the previous column contains two yellow tiles in those two  rows when $i$ is even, or the next column contains two yellow tiles in those two  rows when $i$ is odd.
\end{enumerate}
\end{definition}

When $\al$ is clear from the context, we will use the term {\em tiling} to refer to an $\al$-tiling. 
We consider two tilings to be the same if one is a horizontal translate of the other.  It will follow from the proof of Lemma~\ref{tiling} that all $\al$-tilings are periodic and so can be viewed as lying on a cylinder.  When a tiling is displayed in a figure, we will draw a bounded rectangle and  assume that the two vertical edges  are identified, and indicate with a jagged edge where any black tile crosses this boundary.
See Figure~\ref{uF(4,3,4)} for the four possible $(4,3,4)$-tilings.  
Note that by considering tilings to be on a cylinder, notions such as the number of tiles of each color make sense.  For example, row $1$ of the first tiling  in Figure~\ref{uF(4,3,4)} has four yellow tiles, four black tiles, and a red tile which also intersects row $2$.



In a tiling we will call a square of $R_s$ yellow, red, or black depending on whether the tile covering the square has the corresponding color.  The {\em head} of a red tile is the square it covers in the higher of the two rows.





\begin{figure}
%\begin{wrapfigure}[18]{r}{200pt}
\vspace{-14pt}
\centering
\begin{tikzpicture}[scale =0.5]
\draw(-2,1.5) node{$\uF(4,3,4)=$};
\fill(-1,0) circle(.1);
\fill(0,1) circle(.1);
\fill(1,2) circle(.1);
\fill(2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,0) circle(.1);
\fill(6,1) circle(.1);
\fill(7,2) circle(.1);
\fill(8,3) circle(.1);
\draw (-1,0)--(2,3)--(5,0)--(8,3);
\draw (-1,-.5) node{$x_1$};
\draw (0,.5) node{$x_2$};
\draw (1,1.5) node{$x_3$};
\draw (2,2.5) node{$x_4$};
\draw (3,1.5) node{$x_5$};
\draw (4,.5) node{$x_6$};
\draw (5,-.5) node{$x_7$};
\draw (6.25,.5) node{$x_8$};
\draw (7.25,1.5) node{$x_9$};
\draw (8.25,2.5) node{$x_{10}$};
\end{tikzpicture}

\vs{10pt}

\begin{tikzpicture}[scale=0.4]
\draw[very thin] (0,0) grid (17,-3);
\thirdrow \releft{2} \ro \re{3} \ro \re{3} \ro \blank \re{3} \ro \reright{1}
\secondrow \blank \ro \re{2} \ro \re{2} \ro \re{2} \ro \twobyone \re{2} \ro \re{2} 
\firstrow \twobyone \ro \re{3} \ro \re{3} \ro \re{3} \ro \re{3};
\end{tikzpicture}
\hs{20pt}
\begin{tikzpicture}[scale=.4]
\draw[very thin] (0,0) grid (17,-3);
\thirdrow \re{3} \ro \re{3} \ro \blank \re{3} \ro  \re{3} \ro
\secondrow \blank \ro \re{2} \ro \re{2} \ro \twobyone \re{2} \ro  \re{2} \ro \re{2}
\firstrow \twobyone \ro \re{3} \ro \re{3} \ro \re{3} \ro \re{3}
\end{tikzpicture}

\vs{10pt}

\begin{tikzpicture}[scale=.4]
\draw[very thin] (0,0) grid (17,-3);
\thirdrow \ro \re{3} \ro \blank \re{3} \ro \re{3} \ro  \re{3} 
\secondrow \blank \ro \re{2} \ro \twobyone \re{2} \ro \re{2} \ro  \re{2} \ro \re{2} 
\firstrow \twobyone \ro \re{3} \ro \re{3} \ro \re{3} \ro \re{3}
\end{tikzpicture}
\hs{20pt}
\begin{tikzpicture}[scale=.4]
\draw[very thin] (0,0) grid (5,-3);
\thirdrow \ro \blank \re{3}
\secondrow \ro \twobyone \re{2}
\firstrow \ro \re{3} \twobyone
\end{tikzpicture}
\caption{Antichain orbits in $\uF(4,3,4)$.}
%\caption{ The three antichain orbits of length $17$  and the orbit of length $5$  in $\uF(4,3,4)$.}
\label{uF(4,3,4)}
\end{figure}
%\end{wrapfigure}

Given $\al=(\al_1,\ldots,\al_s)$ we now construct a bijection
$
\phi\colon \{\cO \mid \text{$\cO$ an orbit of $\cA(\al)$}\}
\ra\{T \mid \text{$T$ an $\al$-tiling}\}
$
as follows.  
Let the $i$th segment of $F(\al)$ be $S_i$.
Given $\cO$, we build $T=\phi(\cO)$ column-by-column.  Pick any $A\in\cO$ and any column $C$  of $R_s$ to correspond to $A$.  Color the square in row $i$ of $C$ yellow, red or black depending upon whether $S_i\cap A$ is empty, a shared element, or an unshared element, respectively.  
For example, the antichain $A=\{x_4,x_9\}$ in $\uF(4,3,4)$ corresponds to the first column of the first tiling in Figure~\ref{uF(4,3,4)}.
Now color the column to the right of $C$ in $R_s$ in the same way using the antichain following $A$ in $\cO$, and similarly for the column to the left and the antichain preceding $A$.  Continue this process until all of $R_s$ is colored.  Clearly this is a periodic tiling and so can be wrapped onto a cylinder.  The tilings for the four antichain orbits in $\uF(4,3,4)$ are displayed in Figure~\ref{uF(4,3,4)}.  For example, the tiling with five columns corresponds to the orbit
$
\{
\emp,\ 
\{x_1, x_7\},\ 
\{x_2, x_6, x_8\},\
\{x_3, x_5, x_9\},\
\{x_4, x_{10}\}
\}.
$

\ble
\label{tiling}
For every $\al$, the map $\phi$ is a bijection.
\ele
% \begin{proof}
% We must first prove that $\phi$ is well defined in the sense that if $\cO$ is an orbit of antichains, then $T=\phi(\cO)$ is an $\al$-tiling. To start, we need to show that the colored squares can be partitioned into tiles of the appropriate sizes.  Every individual yellow square is clearly a $1\times 1$ tile.  And the red squares form $2\times 1$ tiles since any shared element of $F=F(\al)$ is in two adjacent segments.  The black tiles will take more work.

% Suppose $x,y$ are unshared elements with $x$ covered by $y$.  We claim that $x\in A$ for some antichain $A$ if and only if $y\in\rho(A)$.  This will ensure  that in row $i$, any set of consecutive black squares will be of length at least $\al_i-1$, since that is the number of unshared elements in the $i$th segment $S_i$.  If $x\in A$, then $y$ is in the complement $U$ of the lower order ideal generated by $A$ since $y\gt x$.  And since $y$ is an unshared cover, $y$ is a minimal element of $U$.  This implies $y\in\rho(A)$.  The reverse implication is similar and left to the reader.

% For the rest of the proof there are two cases depending on whether $i$ is even or odd.  We will just give details for the former, as the other case is very similar. From the previous paragraph, to show that sets of consecutive black squares have length exactly $\al_i-1$, it suffices to prove that, if $x\in A$ is the smallest  (respectively largest) unshared element on $S_i$, then the square before.   Suppose that $x$ is the largest unshared element of $S_i$.  There are two subcases depending on whether $z\in A$ or $z\not\in A$, where $z$ is the largest unshared element of segment $S_{i-1}$.  In the first subcase, the maximal element $S_i\cap S_{i-1}$ is in $\rho(A)$, and so the square after the one for $x$ is covered by a red tile with its head in row $i-1$.  If $z\not\in A$, then $U$ contains an element of $S_{i-1}$ below its maximal element.  It follows that $\rho(A)\cap S_i=\emp$ and so the square after $x$ is a yellow tile.  In much the same way, one sees that if $x$ is the smallest unshared element of $S_i$ then the square before it is covered by a red tile with its head in row $i$ or a yellow tile depending on whether $w\in A$ or $w\not\in A$, where $w$ is the smallest unshared element of $S_{i+1}$.

% We now verify condition (b) in Definition~\ref{TilDef}, postponing condition (a) until the next paragraph.  First suppose the column for $A$ contains a red tile covering rows $i$ and $i+1$.  Since $i$ is even, this corresponds to the minimal element $S_i\cap S_{i+1}$ being in $A$.  But this forces the intersections of $S_i$ and $S_{i+1}$ with $\rho^{-1}(A)$ to be empty.  It follows that the previous column contains yellow squares in those rows as desired.  All these implications are reversible, proving the converse statement.

% Next we verify condition (a). It follows from the two previous paragraphs that any black tile is bounded on the left and on the right  by yellow tiles if one ignores the red ones.  Using similar arguments, one sees that between any two yellow tiles there must be a black tile as long as $\al_i\ge2$, so that black tiles can exist in row $i$.  So (a) follows and we have shown that the map $\phi$ is well defined. 

% To show that $\phi$ is a bijection, we describe  its inverse.  Given an $\al$-tiling, $T$, we construct a family $\cO$ of subsets of $F$ as follows.
% Each column $C$ of $T$ corresponds to an element $A\in\cO$.  
% If the square in row $i$ of $C$ is yellow, then we 
% let $A$ contain none of the elements of $S_i$.  If this square is a part of a red tile which covers row $i-1$ or $i+1$, then we let
% $S_{i-1}\cap S_i$ or $S_i\cap S_{i+1}$ be in $A$, respectively.  Finally, if the square is the $j$th element of a black tile, then we put the $j$th smallest unshared element of $S_i$ in $A$.  Verifying that $A$ is an antichain and that the sequence of antichains forms an orbit does not involve any new ideas and so is left to the reader.  The fact that this is the inverse of $\phi$ is clear since the first part of the proof shows that this is true on the level of the individual columns of the tiling and antichains of the orbit.
% \end{proof}

Using the tiling model it is easy to read off various statistics about orbits which will be useful in proving homomesy and homometry results.  For $x\in F$, consider the {\em indicator function on antichains} $\chi_x\colon \cA(F)\ra\{0,1\}$ defined by
$
\chi_x(A)=\case{1}{if $x\in A$,}{0}{else.}
$
For an orbit $\cO$ of antichains, $\chi_x(\cO)=\sum_{A\in\cO} \chi_x(A)$  is the number of times $x$ occurs in an antichain of the orbit.
We also define
$
\chi(A) = \#A =\sum_{x\in F} \chi_x(A),
$
so that $\chi(\cO)$ is the total number of antichain elements in an orbit.
As usual, we add a hat for the corresponding functions on ideals $I$.  For example, $\chih_x\colon \cI(F)\ra\{0,1\}$ is defined by
$
\chih_x(I)=\begin{cases} 1 & \text{if $x\in I$,} \\ 0 & \text{oterhwise.} \end{cases}
$

To state our first result in this regard we will need some additional notation for fences and tilings.  In $F$, we let
$\uS_i$ be the set of unshared elements of segment $S_i$; 
$s_{i,j}$ be the $j$th smallest element of $\uS_i$; and
$s_i$ be the unique element of $S_i\cap S_{i+1}$.
% \begin{align*} 
% \uS_i &= \text{the set of unshared elements of segment $S_i$,}\\
% s_{i,j}&= \text{the $j$th smallest element of $\uS_i$,}\\
% s_i   &= \text{the unique element of $S_i\cap S_{i+1}$.}
% \end{align*}
Note that we are also using $s$ for the number of segments.  But the presence or absence of a subscript will distinguish between the two uses of this notation.
Note that $s_i$ is a minimal or maximal element of $F$ depending on whether $i$ is even or odd, respectively.   For the fence $\uF(4,3,4)$ in Figure~\ref{uF(4,3,4)} we have: $x_1=s_{1,1}$, $x_2=s_{1,2}$, $x_3=s_{1,3}$, $x_4=s_1$,
$x_5=s_{2,2}$,
$x_6=s_{2,1}$, $x_7=s_2$,
$x_8=s_{3,1}$, $x_9=s_{3,2}$, $x_{10}=s_{3,3}.$

In an $\al$-tiling we let $b_i = $the number of black tiles in row $i$, and 
$r_i =$  the number of red tiles whose head is in row $i$.
% \begin{align*}
% b_i &= \text{the number of black tiles in row $i$,}\\
% r_i &=  \text{the number of red tiles whose head is in row $i$.}\\
% \end{align*}
So, in the first tiling of Figure~\ref{uF(4,3,4)} we get: 
% $$
% \overliner{c|c|c}
% i\   & b_i\   & r_i\ \\
% \hline
% 1   & 4     & 1\\
% 2   & 5     & 1\\
% 3   & 4     & 0
% \earr.
% $$
\begin{small}
\renewcommand{\arraystretch}{0.7}
\begin{tabular}{|l||c|c|c|} \hline
$i$    & 1 & 2 & 3\\
$b_{i}$& 4 & 5 & 4\\ 
$r_{i}$& 1 & 1 & 0 \\ 
\hline 
\end{tabular}\,.
\end{small}

We let $b_i=r_i=0$ if $i\not\in[s]$, where $s$ is the number of rows of the tiling.
Finally, we need the {\em Kronecker function} which, given a statement $R$, evaluates to
\[
\de(R) = \begin{cases} 1 & \text{if $R$ is true,} \\ 0 &\text{if $R$ is false.} \end{cases}
\]

The following result will show how we can use the $\al_i$,  $b_i$ and $r_i$ of the tilings to compute the various quantities in which we are interested.
\begin{lem}
\label{stats}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$ with corresponding fence $F=\uF(\al)$.  Let $\cO$ be an orbit of antichains $A$ with tiling $T=\phi(\cO)$.  We denote by $I$ the ideal generated by $A$.
\begin{enumerate}
    \item[(a)]  For any $i\in[s]$ with $\al_i\ge2$ we have
    $
    \#\cO = b_i\al_i + r_i+r_{i-1}.
    $
    \item[(b)] Considering $\cO$ as an orbit of antichains,
    $
    \chi_x(\cO)=
    \begin{cases}
    b_i &\text{if $x\in \uS_i$,}\\
    r_i &\text{if $x=s_i$},
    \end{cases}
    $ \\ 
and
    $
    \chi(\cO) = \sum_{i=1}^{s} (b_i\al_i-b_i + r_i).
    $
    \item[(c)]  Considering $\cO$ as an orbit of ideals,
    $
    \chih_x(\cO)=
    \begin{cases}
    b_i(\al_i-j)+r_{i-\delta(\text{$i$ even})} &\text{if $x= s_{i,j}$,}\\
    r_{2i-1}&\text{if $x=s_{2i-1}$,}\\
    \#\cO - r_{2i} &\text{if $x=s_{2i}$,}
    \end{cases}
    $\\ 
and
    $
    \chih(\cO) = \left\lfloor \frac{s-1}{2} \right\rfloor \cdot\#\cO
    + \sum_{i=1}^{s} \left[b_i \binom{\al_i}{2}
    +r_{2i-1}(\al_{2i-1}+\al_{2i}-1) -r_{2i} \right].
    $
    % \item[(d)]  Considering $\cO$ as an orbit of antichains,
    % $
    % \chi(\cO) = \sum_{i=1}^{s} (b_i\al_i-b_i + r_i).
    % $
    % \item[(e)]  Considering $\cO$ as an orbit of ideals,    
    % $
    % \chih(\cO) = \left\lfloor \frac{s-1}{2} \right\rfloor \cdot\#\cO
    % + \sum_{i=1}^{s} \left[b_i \binom{\al_i}{2}
    % +r_{2i-1}(\al_{2i-1}+\al_{2i}-1) -r_{2i} \right].
    % $
\end{enumerate}
\end{lem}
\begin{proof}
To illustrate how these results are derived, we will prove (a). Since every black tile in row $i$ has $\al_i-1$ squares, the number of black squares in that row is $b_i(\al_i-1)$.  From Definition~\ref{TilDef} (a), the number of yellow tiles in row $i$ is also $b_i$ as long as $\al_i\ge2$.  Finally, there are red squares in row $i$ from both red tiles whose head is in row $i-1$ and those whose head is in row $i$.  Since the size of the orbit is the number of squares in row $i$, we have
$
\#\cO=b_i(\al_i-1)+b_i+r_{i-1}+r_i,
$
which simplifies to the given quantity.
\end{proof}
% \begin{proof}
% (a)  Since every black tile in row $i$ has $\al_i-1$ squares, the number of black squares in that row is $b_i(\al_i-1)$.  From Definition~\ref{TilDef} (a), the number of yellow tiles in row $i$ is also $b_i$ as long as $\al_i\ge2$.  Finally, there are red squares in row $i$ from both red tiles whose head is in row $i-1$ and those whose head is in row $i$.  Since the size of the orbit is the number of squares in row $i$, we have
% $
% \#\cO=b_i(\al_i-1)+b_i+r_{i-1}+r_i,
% $
% which simplifies to the given quantity.

% \medskip

% (b)  From the proof of Lemma 2.3, we see that if $x\in\uS_i$ then a square of $T$ corresponding to $x$ appears exactly once in each black tile in row $i$.  So the first case follows. 
% On the other hand, the appearances of $x=s_i$, the unique element of $S_i\cap S_{i+1}$, in $\cO$ are in bijection with red tiles covering rows $i$ and $i+1$.  But that is exactly what is counted by $r_i$.

% \medskip

% (c)  An element $x$ will appear in an ideal $I$ if and only if $x\lte y$ for some element $y$ of the antichain of maximial elements of $I$.  So
% \begin{equation}
% \label{chih_x}
%     \chih_x(\cO) = \sum_{y\gte x} \chi_y(\cO).
% \end{equation}


% First consider the case when $x=s_{i,j}$.  Then $y\gte x$ if and only if $y=s_{i,k}$ for $k\ge j$ or $y$ is the maximal element of $S_i$.  And the latter occurs if $y=s_i$ when $i$ is odd or $y=s_{i-1}$ if $i$ is even.  Since $s_{i,k}$ corresponds to the $k$th black square of a black tile, there are $(\al_i-1)-(j-1)=\al_i-j$ elements in a given black tile satisfying the first possibility when $y\gte x$.  And for the second, we have a unique tile with its head in the appropriate row each time $y$ appears.  Combining this with equation~\eqref{chih_x} and part (b) gives the first case of the desired formula.

% When $x$ is shared, it is either a maximal element and thus of the form $x=s_{2i-1}$, or a minimal element and so $x=s_{2i}$ for some $i$.  Maximal elements of $F$ are only in an ideal when they are in the corresponding antichain.  These correspond to the red tiles counted by $r_{2i-1}$.  A minimal element $x=s_{2i}$ will be in every ideal, of which there are $\#O$, except the ones whose antichain contains no elements in either $S_{2i}$ or $S_{2i+1}$.  In $T$ this corresponds to having yellow tiles in a given column in both rows $2i$ and $2i+1$.  But by part (b) of the definition of $\al$-tilings, these pairs of yellow tiles are in bijection with red tiles having their head in row $2i$.  This accounts for the $-r_{2i}$ term in the expression we wished to obtain.

% \medskip

% (d) Using the counts from (b) we obtain
% \begin{align*}
% \chi(\cO)
% &=\sum_{x\in F} \chi_x(\cO)\\
% &=\sum_{i=1}^s \left[\sum_{x\in\uS_i} b_i+\sum_{x=s_i} r_i\right]\\
% &=\sum_{i=1}^s [b_i(\al_i-1)+r_i]
% \end{align*}
% as desired.

% \medskip

% (e) This equation follows from (c) in much the same way that (d) was obtained from (b).  So the details are omitted.
% \end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{General fences}
\label{gf}
We can use Lemma~\ref{stats} to demonstrate various homomesy  results that hold for fences
in general. For example, part (a) of the following result follows immediately from the first equation in Lemma~\ref{stats} (b).
\begin{thm}
\label{homo}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$ with corresponding fence $F=\uF(\al)$ and let $\cO$ be any rowmotion orbit of $F$.
\begin{enumerate}
    \item[(a)] If $x,y\in\uS_i$ for some $i$, then $\chi_x-\chi_y$ is $0$-mesic.
    \item[(b)]  If $x\in\uS_i$, $y=s_i$ and $z=s_{i-1}$, then $\al_i\chi_x+\chi_y+\chi_z$ is $1$-mesic. 
        \item[(c)] If $x=s_{1,j}$ and $y=s_{1,k}$, then $k\chih_x-j\chih_y$ is $(k-j)$-mesic.
    \item[(d)] Let $x=s_{2i-1}$ and $y=s_{2j}$.  If $r_{2i-1}=r_{2j}$ for all orbits $\cO$, then
    $\chih_x+\chih_y$ is $1$-mesic.
    \item[(e)]  If $s$ is odd and all the $\al_i$ are even, then $\#\cO$ is even for all orbits $\cO$.
    \item[(f)]  If $\al_i=2$ for all $i\in[s]$, then $\chi$ is $s/2$-mesic.
\end{enumerate}
\end{thm}
% \begin{proof}
% (a)  By Lemma~\ref{stats} (b) we have
% $\chi_x(\cO)=b_i=\chi_y(\cO)$ and the result follows.

% \medskip

% (b)  Using Lemma~\ref{stats} (a) and (b) we obtain
% $
% \al_i\chi_x(\cO)+\chi_y(\cO)+\chi_z(\cO)
% =\al_i b_i + r_i + r_{i-1} =\#\cO,
% $
% which is equivalent to what we wished to prove.

% \medskip

% (c)  From  Lemma~\ref{stats} (a) and (c) we see that
% \begin{align*}
% k\chih_x(\cO)-j\chih_y(\cO)
% &=k[b_1(\al_1-j)+r_1] - j [b_1(\al_1-k)+r_1]\\
% &=(k-j)[b_1\al_1+r_1]\\
% &=(k-j)\cdot\#\cO.
% \end{align*}

% \medskip

% (d) By Lemma~\ref{stats} (a) and (c) again, as well as the assumption on red tiles,
% $
% \chih_x(\cO)+\chih_y(\cO) = r_{2i-1}+(\#\cO-r_{2j}) =\#\cO.
% $

% \medskip

% (e)  We will calculate the number of squares in the tiling $T=\phi(\cO)$ in two different ways.  Since all the $\al_i$ are even, we have that $\al_i-1\ge 1$ for all $i$.  This implies that there are black tiles in every row.  From Definition~\ref{TilDef} (a) we see that there is a matching between the black tiles and the yellow tiles in row $i$ for all $i$.  And a matched pair covers $\al_i$ squares, which is an even number.  By definition, red tiles cover two squares.  So the total number of squares in $T$ is even.

% But since $T$ lies on an $s\times\#\cO$ cylinder, the number of squares in $T$ is also $s\cdot\#\cO$.  The fact that $s$ is odd forces $\#\cO$ to be even.

% \medskip

% (f)  From Lemma~\ref{stats} (a) and the assumption on $\al$ 
% we have
% $
% 2 b_i + r_i + r_{i-1} =\#\cO
% $
% for all orbits $\cO$ and all $i\in[s]$.  Summing these equations and using Lemma~\ref{stats} (b) gives
% $
% s\cdot\#\cO
% =2\sum_i b_i + \sum_i r_i +\sum_i r_{i-1}
% =2\left(\sum_i b_i + \sum_i r_i\right)
% =2\cdot\chi(\cO)
% $
% from which the desired homomesy follows.
% \end{proof}

\begin{wrapfigure}[6]{r}{240pt}
\vspace{-14pt}
%\bfi
\begin{tikzpicture}[scale = 0.5]
\draw (0,1)--(2,3)--(5,0)--(7,2);
\fill(0,1) circle(.1);
\draw[fill=white](1,2) circle(.1);
\draw[fill=white](2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,0) circle(.1);
\draw[fill=white](6,1) circle(.1);
\draw[fill=white](7,2) circle(.1);
\draw(2,0) node{$I$};
\draw(8,1.5) node{$\mapsto$};
\end{tikzpicture}
\begin{tikzpicture}[scale = 0.5]
\draw (0,1)--(2,3)--(5,0)--(7,2);
\fill(0,1) circle(.1);
\fill(1,2) circle(.1);
\draw[fill=white](2,3) circle(.1);
\draw[fill=white](3,2) circle(.1);
\draw[fill=white](4,1) circle(.1);
\fill(5,0) circle(.1);
\fill(6,1) circle(.1);
\draw[fill=white](7,2) circle(.1);
\draw(2,0) node{$\Ib$};
\end{tikzpicture}
\capt{The ideal complement map \label{Ibar}}
%\efi
\end{wrapfigure}

To state our general result on the $\chih$ statistic in orbits of self-dual posets, we need
some definitions.  
Suppose our poset $P$ is self-dual. In particular, this will be true if $P=F(\al)$ where
$\al$ is a palindrome with an odd number of parts. 
So there is an order-reversing  bijection $\orevbi\colon P\ra P$.  Note that  $I\in\cI(P)$ if and only if $\orevbi(I)\in\cU(P)$.
This permits us to define the {\em ideal complement} (with respect to $\orevbi$) of $I\in\cI(P)$ as
$
\Ib = \orevbi\circ c(I).
$
See Figure~\ref{Ibar} for an example in $\uF(3,3,3)$ where circles are black or white depending on whether they are in the ideal or not, respectively.  The relationship with rowmotion is as follows.
\ble
\label{rho+bar}
Let $P$ be self-dual and fix an order-reversing bijection 
$\orevbi\colon P\ra P$.  Then for all $I\in\cI(P)$ we have
$
\rhoh^{-1}(\Ib) = \ol{\rhoh(I)},
$
where the ideal complements are with respect to $\orevbi$.
\ele
% \begin{proof}
% Since  $\orevbi$ and $c$ commute, they can be applied to get the ideal complement in either order.
% And $\rhoh =  \De^{-1} \circ \na \circ c$,  so it suffices to show that the following diagram commutes:
% $
% \begin{tikzcd}
% I \arrow[r, "c"] 
% & U \arrow[r, "\nabla"] \arrow[d, "\orevbi"] 
% & A \arrow[r, "\Delta^{-1}"] \arrow[d, "\orevbi"] 
% & J \arrow[d, "\orevbi"] &   \\
% & \overline{I} \arrow[r, "\De"]          
% & B \arrow[r, "\nabla^{-1}"]                       
% & V \arrow[r, "c"]   
% & K.
% \end{tikzcd}
% $

% To see that the first square of the diagram commutes, recall that $\orevbi$ is an order-reversing bijection.  Thus it sends the minimal elements of any subset $S\sbe P$ bijectively to the maximal elements of $\orevbi(S)$.  The proof of commutativity for the second square is similar and so omitted.
% \end{proof}

\begin{cor}
\label{DualOrb}
Let $P$ be self-dual with $n=\#P$, and fix an order-reversing bijection $\orevbi\colon P\ra P$.  Let $I\in\cI(P)$.
\begin{enumerate}
 \item[(a)] If  $I,\Ib\in\cO$ for some orbit $\cO$, then
 $
 \frac{\chih(\cO)}{\#\cO}= \frac{n}{2}.
 $
 \item[(b)] If $I\in\cO$ and $\Ib\in\cOb$ for some orbits $\cO$ and $\cOb$  with $\cO\neq\cOb$, then $\#\cO=\#\cOb$ and
 $
 \frac{\chih(\cO)+\chih(\cOb)}{\#\cO+\#\cOb}= \frac{n}{2}.
 $
\end{enumerate}
\end{cor}

\begin{proof}  
We will sketch the proof of (a).  It follows quickly from the definitions that for any ideal $I$ we have $\#I+\#\Ib=n$.  If $\cO$ is an orbit that contains both an ideal and its ideal complement, then one can use the previous lemma to show that the orbit can be partitioned into subsets $\{I,\Ib\}$ where $I\neq \Ib$ and $\{J\}$ where $J=\Jb$.  Since the average of $\chih$ is $n/2$ over each subset, this is also the average over $\cO$.
\end{proof}

% \begin{proof}
% (a)  We claim that the hypothesis on $\cO$ implies that, for any $J\in\cO$, we must have $\Jb\in\cO$.  Indeed, $J=\rhoh^j(I)$ for some integer $j$.  So, by Lemma~\ref{rho+bar},
% $
% \Jb=\ol{\rhoh^j(I)}=\rhoh^{-j}(\Ib)\in\cO,
% $
% since $\Ib\in\cO$.  Thus $\cO$ can be partitioned into pairs $\{I,\Ib\}$ with $I\neq \Ib$, and singletons $\{I\}$ with $I=\Ib$.  In each pair and singleton, the average of $\chih$ is $n/2$, so the same is true of the orbit.

% \medskip

% (b) Suppose $\cO=\{I_1,\ldots,I_k\}$.  Similar reasoning as in (a) shows that the hypothesis on $I,\Ib$ implies that
% $\cOb=\{\Ib_1,\ldots,\Ib_k\}$.  Now the average of $\chih$ in each pair $\{I_i,\Ib_i\}$ is equal to $n/2$, which implies the same is true of $\cO\uplus\cOb$.
% \end{proof}

To turn this last result into a homomesy, take a self-dual $P$ with a given order-reversing bijection  $\orevbi\colon P\ra P$.
Consider the group generated by the action of $\rhoh$ and the ideal complement map $I\mapsto\Ib$.  The orbits of this action will be called {\em superorbits}. 
It can be shown from the proof of Corollary~\ref{DualOrb} that
each superorbit is either an orbit of the action of $\rhoh$ or a union of two such orbits.  
From this observation and the statement of  Corollary~\ref{DualOrb}, the next result follows immediately.
\begin{thm}
\label{n/2-mes}
Let $P$ is self-dual with $n=\#P$ and fix an order-reversing bijection $\orevbi\colon P\ra P$.  Then $\chih$ is $(n/2)$-mesic on superorbits.\hqed
\end{thm}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Fences with few segments}
\label{ffs}

In this section we will consider fences with at most $5$ segments.  For certain compositions $\al$, we will completely describe the orbit sizes and the number of orbits.  We will also calculate the values of $\chi$ and $\chih$, revealing a number of homometries.

We will need an expression for the number of ideals (equivalently, antichains) in
$F=\uF(\al)$, which the following lemma provides. 

\begin{lem}\label{recurrenceI}
Let  $\al=(\al_1,\ldots,\al_s)$ where $s\ge3$.  Then
$
\#\cI(\al)=\al_s\cdot\#\cI(\al_1,\ldots,\al_{s-1})
+\#\cI(\al_1,\ldots,\al_{s-2}).
$
\end{lem}
% \begin{proof}
% We will assume that $s$ is odd; the proof in the case where $s$ even is similar. 
% Let $x$ be the common element of the last two segments which, by the assumption on $s$, is a minimal element of $F=\uF(\al)$.  Then for any $I\in\cI(\al)$ we have $x\in I$ or $x\not\in I$.  Since $x$ is minimal, the ideals with $x\in I$ are in bijection with the ideals of $F-\{x\}\iso\uF(\al_1,\ldots,\al_{s-1})\uplus\uF(\al_s-1)$.  
% Since for any two  posets $P,Q$  we have $\cI(P\uplus Q) =\cI(P)\times \cI(Q)$,
% this accounts for the first term of the recursion.  On the other hand, if $x\not\in I$, then $I$ does not contain any element on either of the last two segments of $F$.  So these ideals contribute the second term, and we are done.
% \end{proof}

\begin{cor}
\label{ideals}
We have the following ideal counts.
\begin{enumerate}
    \item[(a)] If $\al=(a,b)$, then
    $
    \#\cI(\al) = ab+1.
    $
    \item[(b)] If $\al=(a,b,c)$, then
    $
    \#\cI(\al) = abc+a+c.
    $
%%% OMITTING THESE HERE SINCE WE DON'T USE THEM LATER IN THIS ABSTRACT
    % \item[(c)] If $\al=(a,b,c,d)$, then
    % $
    % \#\cI(\al) = abcd+ab+ad+cd+1.
    % $
    %     \item[(d)] If $\al=(a,b,c,d,e)$, then 
    % $ 
    % \#\cI(\al) = abcde+abc+abe+ade+cde+a+c+e.
    % $
\end{enumerate}
\end{cor}

\begin{figure}
\centering
\begin{tikzpicture}[scale=0.4]
\draw[very thin] (0,0) grid (21,-2);
\secondrow \ro \re{3} \ro \re{3} \ro \re{3} \ro \re{3} \ro \re{3} \blank
\firstrow \ro \re{4} \ro \re{4} \ro \re{4} \ro \re{4} \twobyone;
\end{tikzpicture}
\caption{The orbit $\cO$ of length $21$ in $\uF(5,4)$, with $\chi(\cO)=32$ antichain elements.}
\label{uF(5,4)}
\end{figure}


We start with the case of two segments.  
The following result can be proved using the tiling counts in Lemma~\ref{stats}.
Note that parts (c) and (d) are homometries which are not homomesies.
\begin{thm}
\label{uF(a,b)}
Rowmotion on $\uF(a,b)$ has the following properties. 
\begin{enumerate}
\item[(a)] All orbits $\cO$ have size 
$\ell=\lcm(a,b)$
except for one $\cO'$ which has size $\ell+1$.
\item[(b)] The number of orbits is $\gcd(a,b)$.
\item[(c)] For a size $\ell$ orbit, 
$
\chi(\cO)=\frac{2ab-a-b}{\gcd(a,b)}:=m.
$
For the size $\small \ell+1$ orbit, 
$
\chi(\cO') = m+1.
$
\item[(d)]  For a size $\ell$ orbit, 
$
\chih(\cO)=\frac{\ell(a+b-2)}{2}.
$
For the size $\small \ell+1$ orbit, 
$
\chih(\cO')=\frac{(\ell+2)(a+b-2)}{2}+1.
$
\end{enumerate}
\end{thm}
% \begin{proof}
% (a)  There is only one antichain containing the unique shared element $s_1$ of $\uF(a,b)$.  So there is a unique tiling $T'$ with a red tile and all other tilings $T$ consist only of yellow and black tiles. Figure~\ref{uF(5,4)} displays $T'$ for $\uF(5,4)$.  

% Consider a tiling $T$ of the second type.  From Definition~\ref{TilDef} (a), the first row can be partitioned into pairs consisting of a black tile and a yellow tile which together cover $a$ squares.  Similarly the second  row has a partition into blocks of $b$ squares.  So $T$ must have $\ell=\lcm(a,b)$ columns.  In the same manner we see that $T'$ must have one more column for the red tile.  This implies the desired orbit lengths.

% \medskip

% (b)  Let $k$ be the number of orbits.  Using part (a) and Corollary~\ref{ideals},
% $
% ab+1=\#\cI(a,b) = (k-1)\ell+(\ell+1)=k\ell+1.
% $
% It follows that $k=\gcd(a,b)$.

% \medskip

% (c)  From the description of the tilings in (a), for all orbits we have
% $
% b_1=\frac{\lcm(a,b)}{a}=\frac{b}{\gcd(a,b)}.
% $
% Similarly $b_2=a/\gcd(a,b)$.  Finally $r_1=0$ or $1$ for the orbits $\cO$ and $\cO'$, respectively.  The result now follows by substitution into Lemma~\ref{stats} (d), using that $\al_1=a$ and $\al_2=b$. 

% \medskip

% (d)  These equations follow from (a) and Lemma~\ref{stats} (e) in much the same way as the derivation in part (c).  So we leave the details to the reader.
% \end{proof}

In the next result we consider certain fences with three segments.  We do not explicitly
state the values of the $\chi$ statistic  for the homometries, but these can be computed
from the numbers of tiles of each color, which are given in the proof in~\cite{EPRS:rf}. 

\begin{thm}
\label{uF(a,b,a)}
Consider $\uF(a,b,a)$  and define
$
g=\gcd(a,b),\quad
\ab = a/g,\quad
\bba = b/g,\quad
\ell=\lcm(a,b).
$
Since $\ab,\bba$ are relatively prime, there exists a smallest positive integer $m$ such that
$
m\ab = q \bba +1
$
for some positive integer $q$.  Then the orbits of rowmotion on $\uF(a,b,a)$ can be partitioned by length into three sets $\cS,\cM,\cL$, which we call {\em small}, {\em medium}, and {\em large}, having the following properties:
\begin{enumerate}
    \item[(a)] We have
    $
    \#\cO =
    \begin{cases}
    \ell&\text{if $\cO\in\cS$},\\
    a(2b-2\bba+m)+g&\text{if $\cO\in\cM$},\\
    a(2b-\bba+m)+g&\text{if $\cO\in\cL$},
    \end{cases}
    $ \\
with
    $
    \#\cS = \ab(g-1)^2,\quad
    \#\cM = \frac{\ab m -1}{\bba\rule{0pt}{10pt}},\quad
    \#\cL = \frac{\ab (\bba- m) +1}{\bba\rule{0pt}{10pt}}.
    $
    \item[(b)] For rowmotion on antichains, $\chi$ is homometric. 
    \item[(c)] For rowmotion on ideals, $\chih$ is $n/2$-mesic where $n=\#\uF(a,b,a)=2a+b-1$.
\end{enumerate}
\end{thm}
% \begin{proof}
% (a) We will construct the corresponding tilings, where we will number the columns from left to right starting with column $0$.  We can use any integer to refer to a column by using its standard representative modulo $\#\cO$, the number of columns.
% Figure~\ref{uF(4,3,4)} gives an example of the tilings for $\uF(4,3,4)$, where $\cS=\emp$ because $g=1$.

% We first construct the tilings $T$ for the orbits in $\cS$. 
% We also standardize these tilings so that the first row begins with a yellow tile.
% The first row of $T$ will consist of $\ell/a$ copies of a pair consisting of a yellow tile followed by a black tile of length $a-1$.  So $T$ has length $\ell$.
% We now tile the second row with $\ell/b$ copies of a paired yellow tile and black tile of length $b-1$ positioned in such a way that no two yellow tiles from the first two rows are in the same column.  Finally tile the third row with alternating yellow and length $a-1$ black tiles, requiring that no two yellow tiles from the last two rows are in the same column.  Clearly this is a valid tiling of length $\ell$.  So, for this case, it remains to calculate the number of rotationally different tilings obtained.

% The first square of a black tile in the second row could be placed under any one of the $a$  squares of the union of a yellow tile and an adjacent black tile in the first row.  Since there are $\ell/b$ black tiles in the second row, any placement of the first black tile will be rotationally equivalent to that many other placements.  So the number of nonequivalent placements is $a/(\ell/b)=g$.  But one of these will be the rotation class which has two yellow tiles in the same column. Thus the count for the placement of the second row is really $g-1$.  Similarly one sees that the number of inequivalent placements for the third row is $\ab(g-1)$, giving $\ab(g-1)^2$ as the final total.

% We will now construct $\ab$ tilings  $T_1,\ldots,T_{\ab}$,
% where the $T_k$ for $k\le (\ab m-1)/b$ will have $a(2b-2\bba+m)+g$ columns and correspond to orbits in $\cM$, and those for the remaining values of $k$ will have $a(2b-\bba+m)+g$ columns and correspond to orbits in $\cL$.  All these tilings will have $g$ red tiles with their heads in row $1$ and $g$ red tiles with their heads in row $2$.  For all $k$, the former tiles of $T_k$ will be in columns
% $
% 0,\ \ell+1,\ 2(\ell+1),\ \ldots,\ (g-1)(\ell+1),
% $
% and the latter will be in columns
% $
% -kb,\ -kb-(\ell+1),\ -kb-2(\ell+1),\ \ldots,\
% -kb-(g-1)(\ell+1),
% $
% where the indices are taken modulo the number of columns  of the tiling.  It is now a matter of simple arithmetic to check that these placements of red tiles can be (uniquely) extended to full tilings.  Also, these tilings are distinct, since the red tiles with heads in the first row of all $T_k$ are fixed in this description, while the red tiles with heads in the second row vary.

% Finally we must verify that these tilings correspond to all orbits by adding up the orbit sizes and making sure they sum to $a(ab+2)$, which is the total number of antichains in $\uF(a,b,a)$ by Corollary~\ref{ideals} (b).  This is a simple, if tedious, computation and is left to the reader.

% \medskip

% (b)  We see from their descriptions in part (a) that these tilings have the following parameters.
% $
% \begin{array}{c|ccc}
% \text{type of orbit}
%         & b_1=b_3   & b_2       & r_1=r_2\\
% \hline
% \cS     &\bba        &\ab    &0\rule{0pt}{12pt}\\
% \cM     &2b-2\bba+m  &2a-2\ab+q  &g\\
% \cL     &2b-\bba+m   &2a-\ab+q   &g
% \end{array}
% $
% Since $\chi$ only depends on these parameters by Lemma~\ref{stats} (d), and they are constant on orbit sizes, this statistic is homometric.

% \medskip

% (c)  For the orbits  $\cO\in\cS$, using Lemma~\ref{stats} (e) and the previous table, we obtain
% $
% \chih(\cO)
% =\ell+2 \bba\binom{a}{2}+\ab\binom{b}{2}
% =\ell+\ell(a-1)+\frac{\ell(b-1)}{2}
% =\ell\cdot \frac{2a+b-1}{2}
% =\ell\cdot \frac{n}{2}
% $
% as desired.

% For the orbits in $\cO\in\cM\cup\cL$ we will use Corollary~\ref{DualOrb} (a).  It suffices to find an ideal $I\in\cO$ with $\overline{I}\in\cO$.  Let $I$ be the ideal corresponding to the zeroth column of the corresponding tiling $T$.  We claim that the ideal $J$ corresponding to column $-kb-1$ satisfies $J=\overline{I}$.  By the choice of $I$, it must contain $s_1$, the intersection of the first two segments.
% Suppose $I$'s maximum element on the third segment is the $t$th unshared element from the bottom, where $t=0$ if $I$ contains no such element.

% The column for $J$ comes just before the red tile in column $-kb$.  So $J$ can have no elements in either the second or third segments.  Furthermore, in the first row there is an alternating sequence of black and yellow tiles, starting with a black tile whose last square is in the last column and working backwards, because of the red tile in column $0$.  Similarly, the red tile in column $-kb$ guarantees a similar sequence of the same length starting in column $-kb+1$ and going forwards in the third row.  It follows that $J$ has the $(a-t-1)$st unshared element on segment one as its maximal element.  But this matches the description of $\overline{I}$ given by applying $c$ and then applying $\orevbi$.
% \end{proof}

% \begin{figure}
% \centering
% \begin{tikzpicture}[scale=0.4]
% \draw[very thin] (0,0) grid (3,-4);
% \firstrow \ro \re{2}
% \secondrow \releft{1} \ro \reright{1}
% \thirdrow \ro \re{2}
% \fourthrow \re{2} \ro

% \begin{scope}[shift={(6,0)}]
% \draw[very thin] (0,0) grid (4,-4);
% \firstrow \ro \re{2} \twobyone
% \secondrow \ro \re{2} \blank
% \thirdrow \releft{1} \ro \twobyone \reright{1}
% \fourthrow \releft{1} \ro \blank \reright{1}
% \end{scope}

% \begin{scope}[shift={(13,0)}]
% \draw[very thin] (0,0) grid (13,-4);
% \firstrow \ro \re{2} \ro \re{2}\ro \re{2}\ro \re{2} \twobyone
% \secondrow \ro \twobyone \re{2} \ro \twobyone \re{2} \ro \twobyone \re{2} \blank
% \thirdrow \ro \blank \re{2} \ro \blank \re{2} \ro \blank \re{2} \twobyone
% \fourthrow \ro \re{2} \ro \re{2}\ro \re{2}\ro \re{2} \blank
% \end{scope}

% \begin{scope}[shift={(-2,-5)}]
% \draw[very thin] (0,0) grid (30,-4);
% \firstrow \ro\re{2} \ro\re{2} \ro\re{2} \ro\re{2} \ro\re{2} \ro\re{2} \ro\re{2}\twobyone \ro\re{2}\twobyone \ro\re{2}\twobyone
% \secondrow \ro\twobyone\re{2} \ro\twobyone\re{2} \ro\re{2} \ro\re{2} \ro\re{2} \ro\twobyone\re{2}\blank \ro\re{2}\blank \ro\re{2}\blank
% \thirdrow \ro\blank\re{2} \ro\blank\re{2} \twobyone\ro\re{2} \twobyone\ro\re{2} \twobyone\ro\blank\re{2} \ro\re{2} \ro\re{2} \ro\re{2}
% \fourthrow \re{2} \ro\re{2} \ro\re{2}\blank \ro\re{2}\blank \ro\re{2}\blank \ro\re{2} \ro\re{2} \ro\re{2} \ro\re{2} \ro
% \end{scope}
% \end{tikzpicture}
% \capt{An orbit of $\uF(3,3,3,3)$ in each of the four sizes.}
% \label{uF(3,3,3,3)}
% \end{figure}



% We next consider the case of a $4$-segment fence with all parts of $\al$ equal.  An example of the orbits is given  in Figure~\ref{uF(3,3,3,3)}.


% \begin{thm}
% \label{uF(a,a,a,a)}
% The orbits of rowmotion on $\uF(a,a,a,a)$ can be partitioned into four sets $\cS,\cM,\cL,\cG$ called
% {\em small}, {\em medium}, {\em large}, and {\em gigantic} having the following properties.
% \begin{enumerate}
%     \item[(a)] 
%     $
%     \#\cS = (a-1)^3,\quad
%     \#\cM = a,\quad
%     \#\cL = 1,\quad
%     \#\cG = a-1,
%     $
%     and
%     $
%     \#\cO =
%     \begin{cases}
%     a&\text{if $\cO\in\cS$},\\
%     a+1&\text{if $\cO\in\cM$},\\
%     a^2+a+1&\text{if $\cO\in\cL$},\\
%     3a^2+a&\text{if $\cO\in\cG$}.
%     \end{cases}
%     $
%     \item[(b)]  The statistics $\chi$ and $\chih$ are homometric under rowmotion, with values given in the following chart:
%     $
%     \overliner{c|cccc}
%             &\cS    &\cM    &\cL    &\cG\\
%     \hline
%     \chi(\cO)& 4a-4 & 4a-2  & 4a^2-a& 12a^2-11a+2\\
%     \chih(\cO)&2a^2-a&2a^2+3a-1&2a^3+3a-1&6a^3-a
%     \earr
%     $
% \end{enumerate}
% \end{thm}
% \begin{proof}
% (a) We will construct the corresponding tilings, labeling  the columns starting at 0. For $1\le i\le 4$, let $j_i$ denote the smallest nonnegative number so that row $i$ of the tiling has a yellow square in column $j_i$. 
% We set $j_1=0$, since one can always choose the first antichain in the orbit to have a yellow square in the top row. Let us consider four types of orbits as follows.


% \medskip

% $\cS$: $0\neq j_2\neq j_3\neq j_4$. In such an orbit, each row simply consists of one yellow and one black tile, and the orbit has length $a$. For $2\le i\le 4$, there are $a-1$ possibilities for $j_i$, since it can be any integer between $1$ and $a$ other than $j_{i-1}$. Thus, there are $(a-1)^3$ such orbits. 
% \medskip

% $\cM$: $0= j_2\neq j_3= j_4$. In such an orbit, rows 1 and 2 both consist of a yellow tile in column 0, followed by a black tile, followed by a red tile with its head in row 1. Rows 3 and 4 have the same configuration, shifted $j_3$ positions to the right. Therefore, such an orbit has length $a+1$. Since $j_3$ can be any integer between $1$ and $a$, there are $a$ such orbits. 
% \medskip

% $\cL$: $0= j_2= j_3= j_4$. We will construct a unique orbit of this type where every row intersects at least one red tile.  So it suffices to specify the positions of these tiles. There are two red tiles in column $a(a+1)$ (the rightmost column), with heads in rows 1 and 3. Additionally, there are $a$ red tiles with their head in row 2, in columns $1+r(a+1)$ for $0\le r\le a-1$. It is straightforward to check that these red tiles can be extended to a full tiling.
 
% \medskip

% $\cG$: $0= j_2= j_3\neq j_4$. Let us write $j$ instead of $j_4$ for short. 
% For each value of $j$ satisfying $1\le j\le a-1$, we construct an orbit of length $3a^2+a$.
% Again, it suffices to specify the positions of the red tiles, since every row intersects at least one red tile. From left to right, the red tiles are in the following positions:
% \begin{itemize}
%     \item In columns $1+r(a+1)$ for $0\le r\le j-1$, with head in row 2.
%     \item In columns $r(a+1)$ for $j\le r\le a+j-1$, with head in row 3.
%     \item In columns $2+r(a+1)$ for $a+j-1\le r\le 2a-2$, with head in row 2.
%     \item In columns $1+r(a+1)$ for $2a-1\le r\le 3a-2$, with head in row 1.
% \end{itemize}

% It can be checked that this configuration of red tiles can be extended to a valid tiling. Note  that the $a-1$ orbits obtained for the different values of $j$ are all distinct. This is because the positions of the red tiles with heads in row $1$ are independent of $j$, whereas the sets of positions of the other red tiles are different for each $j$.

% Next we show that we have obtained tilings for all the orbits. The total number of antichains in the orbits that we have constructed is
% $
% (a-1)^3\cdot a+ a\cdot (a+1)+(a^2+a+1)+(a-1)\cdot(3a^2+a)=a^4+3a^2+1,
% $
% which equals $\#\cI(a,a,a,a)$ by Corollary~\ref{ideals} (c).


% \medskip

% (b) From the constructions in (a) we see that, for given $i$, the paramenters $b_i$ and $r_i$ are constant in each of the four classes of orbits.  Orbomesy follows immediately.  To compute the statistics, we note the following specific values.
% $
% \begin{array}{c|cccc}
% \text{type of orbit}
%         & b_1=b_4   & b_2=b_3   & r_1=r_3   & r_2\\
% \hline
% \cS     & 1         & 1         & 0         & 0\rule{0pt}{12pt}\\
% \cM     & 1         & 1         & 1         & 0\\
% \cL     & a+1       & a         & 1         & a\\
% \cG     & 3a        & 3a-1      & a         &a
% \end{array}
% $
% Now the formulas in Lemma~\ref{stats} (d) and (e) complete the proof.
% \end{proof}

In~\cite[]{EPRS:rf} we give similar results for 4-segment fences $\uF(a,a,a,a)$ 
and for 5-segment fences $\uF(a,1,a,1,a)$. 

% We end this section by considering certain fences with $5$ segments.

% \begin{thm}
% \label{uF(a,1,a,1,a)}
% The orbits of rowmotion on $\uF(a,1,a,1,a)$  can be partitioned into 
% three sets $\cS,\cM,\cL$  called {\em small}, {\em medium}, and {\em large} having the following properties.
% \begin{enumerate}
%     \item[(a)] We have
%     $
%     \#\cS = 2a-2,\quad
%     \#\cM = 1,\quad
%     \#\cL = a,
%     $
%     and
%     $
%     \#\cO =
%     \begin{cases}
%     a+1&\text{if $\cO\in\cS$},\\
%     3a+2&\text{if $\cO\in\cM$},\\
%     a^2+2a&\text{if $\cO\in\cL$}.
%     \end{cases}
%     $
%     \item[(b)] For on antichains, $\chi$ is homometric  with
%     $
%     \chi(\cO) =
%     \begin{cases}
%     3a&\text{if $\cO\in\cS$},\\
%     9a-3&\text{if $\cO\in\cM$},\\
%     3a^2+3a-2&\text{if $\cO\in\cL$}.
%     \end{cases}
%     $   
%     \item[(c)] The superorbits are pairs of orbits in $\cS$ and single orbits in $\cM$ and $\cL$.
% \end{enumerate}
% \end{thm}
% \begin{proof}
% (a)
% We label the columns of a tiling as usual.  Since there will be no black tiles in rows 2 and 4, every row will intersect at least one red tile.  Thus it suffices to specify the positions of these tiles.  We will omit the straightforward check that the given positions of the red tiles can be extended to a full tiling with no column repeated, and that the orbits are distinct and have a total of  $a^3+4a^2+3a$ columns, which is the total number of antichains of the fence by Corollary~\ref{ideals} (d).  We label the orbits in $\cL$ by $\cO_k$ where $1\le k\le a$.  We label some of the orbits  in $\cS$ by $\cO_k'$ and the remaining ones by $\cO_k''$  where $1\le k\le a-1$.  Finally, the sole orbit in $\cM$ will be labeled $\cO'$.

% We now give the column numbers for the red tiles in each orbit with their head in each row.  The orbits $\cO_k$ have $r_i=a$ for $1\le i\le 4$, and so the positions will be parameterized by $q$ where $0\le q \le a-1$.  A dash in the following table indicates that there are no red tiles for that orbit with their head in the corresponding row.
% $$
% \begin{array}{l||c|c|c|c}
% \text{orbit}  &
% \cO_k & \cO' & \cO_k' & \cO_k''\\
% \hline
% \text{row $1$}&
% q(a+1)          & 0, a+1    & 0     & 0\\
% \text{row $2$}& 
% q(a+2)+2        & -a        & -     & k+1\\
% \text{row $3$}& 
% q(a+2)          & a         & -1    & -\\
% \text{row $4$}& 
% (q-k+1)(a+1)-k  & 0, -a-1   & k     & k\\
% \end{array}
% $$

% (b)  The following chart of the $r_i$ and $b_i$ values reduces this proof to a routine computation.
% $$
% \begin{array}{l||c|c|c|c|c|c|c|c}
% \text{orbit}  &
% \multicolumn{2}{c|}{\cO_k} & 
% \multicolumn{2}{c|}{\cO'} & 
% \multicolumn{2}{c|}{\cO_k'} & 
% \multicolumn{2}{c}{\cO_k''}\\
% & b_i & r_i & b_i & r_i & b_i & r_i & b_i & r_i\\ 
% \hline
% \text{row $1$}&
%   a+1   & a &  3  &  2  &  1  &  1  &  1  &  1\\
% \text{row $2$}& 
%   0   & a   &  0  &  1  &  0  &  0  &  0  &  1\\
% \text{row $3$}& 
%   a   & a   &  3  &  1  &  1  &  1  &  1  &  0\\
% \text{row $4$}& 
%   0   & a   &  0  &  2  &  0  &  1  &  0  &  1  \\
% \text{row $5$}&
%   a+1   & 0 &  3  &  0  &  1  &  0  &  1  &  0  
% \end{array}
% $$


% (c)  First of all, the lone orbit of size $3a+2$ must also be a superorbit  since it has no other orbit with which it could be paired.  To show that the orbits of size $a^2+2a$ are also superorbits, it suffices to find an ideal  $I\in \cO_k$ with $\overline{I}=\orevbi(I)\in\cO_k$.  We take as $I$ the ideal corresponding to column $0$, which consists of red tiles with their heads in rows $1$ and $3$, and the cell in row $5$ covered by the $k$th cell of a black tile for $1\le k\le a-1$ or by a yellow tile for $k=a$.  One can now verify that $\overline{I}$ corresponds to column $(a-k-1)(a+2)+1$ of $\cO_k$.

% Finally, we show that $\orevbi$ pairs $\cO_k'$
% with $\cO_k''$.  Let $I$ correspond to column $a$ in $\cO_k'$.  This column consists of the last cell of a black tile in row $1$, a yellow tile in row $2$, a red tile in rows $3$ and $4$, and the $(a-1)$st cell of a black tile in row $5$. It is not hard to see that $\overline{I}$ corresponds to column in $\cO_k''$ which contains a red tile with its head in row 4.
% \end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Toggling}
\label{tog}
Recall that the rowmotion map can be written as a Coxeter element in the ideal toggle group
$\mathcal{T}$ (Theorem~\ref{thm:rowcox}).  A natural question is whether other toggling orders
might change the homomesies or homometries of the map.  In this section we show that any
map defined as a Coxeter element in the toggle group of certain finite posets $P$ (in particular,
any fence) has the same
homomesies and homometries as any other.  
Our theorem generalizes \cite[Theorem 20]{JR:tispg}, which proves a similar  result in a different context, but one that is essentially the same as order-ideal toggling in zigzag posets which are fences of the form $\uF(2,1,1,\ldots,1,2)$.
The crucial concept in determining which toggle groups have this property is a graph associated with the group.



Let $S=\{x_1,x_2,\ldots,x_n\}$ be a set and let $\cT$ be the toggle group associated with some family  of subsets of $S$.   The {\em base graph} of $\cT$, denoted $G_\cT$, has as vertices the toggles $\tau_x$ for $x\in S$, and it has an edge $\tau_x\tau_y$ if  $\tau_x$ and $\tau_y$ do not commute.  
If $P$ is a poset, we let $G_P=G_{\cT_P}$ and $\Gh_P=G_{\cTh_P}$ be the graphs for the toggle groups for $\cA(P)$ and $\cI(P)$, respectively. 
We will reserve $\tau_i$ for the toggle with respect to $x_i$ in $\cA(P)$ and use $\tauh_i$ for the corresponding toggle in $\cI(P)$, and similarly for other notation involving fences.

%\bfi
\begin{wrapfigure}[11]{r}{220pt}\centering
\begin{tikzpicture}[scale = 0.5,baseline]
\fill(1,2) circle(.1);
\fill(2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,2) circle(.1);
\draw (1,2)--(2,3)--(4,1)--(5,2);
\draw (1,1.5) node{$x_1$};
\draw (2,2.5) node{$x_2$};
\draw (3,1.5) node{$x_3$};
\draw (4,.5) node{$x_4$};
\draw (5,1.5) node{$x_5$};
\draw(3,-0.25) node{$F=\uF(2,2,2)$};
\end{tikzpicture}
\qquad
\begin{tikzpicture}[scale = 0.5,baseline]
\fill(1,2) circle(.1);
\fill(2,2) circle(.1);
\fill(3,2) circle(.1);
\fill(4,2) circle(.1);
\fill(5,2) circle(.1);
\draw (1,2)--(5,2);
\draw (1,1.5) node{$\tau_1$};
\draw (2,1.5) node{$\tau_2$};
\draw (3,1.5) node{$\tau_3$};
\draw (4,1.5) node{$\tau_4$};
\draw (5,1.5) node{$\tau_5$};
\draw (4,2) arc (0:180:1);
\draw(3,-0.25) node{$G_F$};
\end{tikzpicture}\vs{10pt}

\begin{tikzpicture}[scale = 0.5,baseline]
\fill(1,2) circle(.1);
\fill(2,2) circle(.1);
\fill(3,2) circle(.1);
\fill(4,2) circle(.1);
\fill(5,2) circle(.1);
\draw (1,2)--(5,2);
\draw (1,1.5) node{$\tauh_1$};
\draw (2,1.5) node{$\tauh_2$};
\draw (3,1.5) node{$\tauh_3$};
\draw (4,1.5) node{$\tauh_4$};
\draw (5,1.5) node{$\tauh_5$};
\draw(3,-0.25) node{$\Gh_F$};
\end{tikzpicture}
\qquad
\begin{tikzpicture}[scale = 0.5,baseline]
\fill(1,2) circle(.1);
\fill(2,2) circle(.1);
\fill(3,2) circle(.1);
\fill(4,2) circle(.1);
\fill(5,2) circle(.1);
\draw (1,2)--(5,2);
\draw[->] (1.4,2)--(1.6,2);
\draw[->] (2.4,2)--(2.6,2);
\draw[->] (3.6,2)--(3.4,2);
\draw[->] (4.6,2)--(4.4,2);
\draw (1,1.5) node{$\tauh_1$};
\draw (2,1.5) node{$\tauh_2$};
\draw (3,1.5) node{$\tauh_3$};
\draw (4,1.5) node{$\tauh_4$};
\draw (5,1.5) node{$\tauh_5$};
\draw(3,-0.25) node{$O(\tauh_1\tauh_5\tauh_2\tauh_4\tauh_3$)};
\end{tikzpicture}
\capt{The fence $F=\uF(2,2,2)$, its\\\hspace{\textwidth} graphs $G_F$ and $\Gh_F$, and an orientation. \label{graphs}}
%\efi
\end{wrapfigure}

Figure~\ref{graphs} displays the fence $F=\uF(2,2,2)$ as well as $G_F$ and $\Gh_F$.  We will need another result of Cameron and Fon-Der-Flaass.
% \bth[\cite{CF:oar}]
% \label{AcyP}
% If $P$ is a finite poset and $x,y\in P$ are such that neither element covers the other, then $\tauh_x$ and $\tauh_y$ commute.  In particular, if the Hasse diagram of $P$ (considered as a graph) is acyclic, then so is $\Gh_P$.\hqed
% \eth
\bth[\cite{CF:oar}]
\label{AcyP}
If $P$ is a finite poset then $\Gh_P$ is the Hasse diagram of $P$ considered as an
undirected graph, \textit{i.e.}, $\tauh_x$ and $\tauh_y$ commute if and only if $x$ and $y$ are not
related by a cover.  In particular, if the Hasse diagram of $P$ (considered as a graph) is acyclic, then so is $\Gh_P$.\hqed
\eth

The acyclicity of $G_\cT$ has consequences for the corresponding Coxeter elements.  In particular, in this case any two Coxeter elements can be connected by a sequence of conjugations of a certain type.  Let $\cT$ be a toggle group and $w$ a Coxeter element of $\cT$. We call $\tau_x$ {\em admissible} with respect to $w$ if all the elements to one side (either left or right) of $\tau_x$ in $w$ commute with $\tau_x$.  If $\tau_x$ is admissible with respect to $w$, then the conjugate $\tau_x w \tau_x$ is again a Coxeter element.  Indeed, suppose all elements to the left of $\tau_x$ in $w$ commute with $\tau_x$.  Then $\tau_x$ can be moved via commutation relations to the left end of $w$ and so cancel in the conjugation.  Continuing the example in Figure~\ref{graphs}, if we let $w=\tauh_1\tauh_5\tauh_2\tauh_4\tauh_3$, then $\tauh_5$ is admissible, since the only element to its left is $\tauh_1$, which commutes with $\tauh_5$.  So\\ 
% \begin{align*}
% \tauh_5 w \tauh_5
% &= \tauh_5(\tauh_1\tauh_5\tauh_2\tauh_4\tauh_3 ) \tauh_5\\
% &=\tauh_5(\tauh_5\tauh_1\tauh_2\tauh_4\tauh_3 ) \tauh_5\\
% &=\tauh_1\tauh_2\tauh_4\tauh_3\tauh_5,
% \end{align*}
$\tauh_5 w \tauh_5
= \tauh_5(\tauh_1\tauh_5\tauh_2\tauh_4\tauh_3 ) \tauh_5
=\tauh_5(\tauh_5\tauh_1\tauh_2\tauh_4\tauh_3 ) \tauh_5
=\tauh_1\tauh_2\tauh_4\tauh_3\tauh_5$, which is still a Coxeter element.

One can relate admissible toggles to the base graph as follows.  Given a Coxeter element $w$, we will form an orientation $O(w)$ of $G_\cT$.  Given an edge in $G_\cT$ between two toggles, we orient it from the toggle which is further left in $w$ to the one which is further right.
The orientation for $\tauh_1\tauh_5\tauh_2\tauh_4\tauh_3$ is shown in Figure~\ref{graphs}.  It is now easy to see that $\tau_x$ is admissible with respect to $w$ if and only if $\tau_x$ is either a source or a sink in $O(w)$.  Indeed, the first case corresponds to all elements to the left of $\tau_x$ commuting with the toggle and the second to this property on the right.  When the base graph is acyclic, then conjugation by admissible toggles is all that is needed to get between any two Coxeter elements, as proved by Eriksson and Eriksson.
\begin{thm}[\cite{EE:cce}]
\label{AcyG}
Let $\cT$ be a toggle group with $G_\cT$ acyclic, and let $w,w'$ be any two Coxeter elements of $\cT$.  Then $w'$ can be obtained from $w$ by a sequence of conjugations where at each step the conjugating toggle is admissible with respect to the current Coxeter element.\hqed
\end{thm}
From all this follows the main result of this section.
\begin{thm}
Let $S$ be a finite set and $\cT$ be a toggle group on $S$ with $G_\cT$ acyclic.  Let $w,w'$ be any two Coxeter elements of $\cT$, and let $W, W'$ be the respective groups they generate.  Then any linear combination of indicator functions $\chi_y$ for $y\in S$ is $d$-mesic 
or homometric
under the action of $W$ if and only if it is $d$-mesic
or homometric, respectively,
under the action of $W'$.
\end{thm}
% \begin{proof}
% By the previous theorem, it suffices to show that the conclusion holds when 
% $w'=\tau_x w \tau_x$ where $\tau_x$ is admissible with respect to $w$.
% To this end, we just need  to show that there is a bijection between the orbits $\cO$ of $W$ and the orbits $\cO'$ of $W'$ such that, for any $y\in S$ and any $\cO\leftrightarrow\cO'$, we have $\chi_y(\cO)=\chi_y(\cO')$.  

% Let $\cO=\{S_1,S_2,\ldots,S_k\}$ where $S_{i+1} = w S_i$ for all $i$ modulo $k$.  We claim that $\cO'=\{S_1',S_2',\ldots,S_k'\}$ is an orbit of $W'$ where $S_i'=\tau_x S_i$.  This follows since
% $$
% w' S_i' = (\tau_x w \tau_x)(\tau_x S_i)
% =\tau_x w S_i = \tau_x S_{i+1}=S_{i+1}'.
% $$

% The theorem will hold for linear combinations of indicator functions if it holds for each individual indicator $\chi_y$.  First consider the case when 
% $y\neq x$.  Then toggling $x$ does not change whether $y$ is in a given subset.  So for any $i$ we have
% $$
% \chi_y(S_i') = \chi_y(\tau_x S_i) = \chi_y S_i.
% $$
% This implies $\chi_y(\cO')=\chi_y(\cO)$ as desired.

% Finally, consider the case when $y=x$ and write $w=w_1\tau_x w_2$.  
% Since $x$ is admissible with respect to $w$, we can move $\tau_x$ to the beginning or end of $w$.  We will do the case when we can write $w=\tau_x w_1 w_2$, as the other is similar.  But now
% $$
% \chi_x(S_i') 
% =\chi_x(\tau_x S_i)
% =\chi_x(\tau_x w S_{i-1})
% =\chi_x(w_1 w_2 S_{i-1})
% =\chi_x(S_{i-1}),
% $$
% where the last step is because none of the toggles in $w_1w_2$ involve $x$.  This sets up an 
% indicator-preserving bijection 
% $S_i'\leftrightarrow S_{i-1}$ between $\cO'$ and $\cO$, so we are done.
% \end{proof}

Combining the previous theorem with Theorem~\ref{AcyP}, we get the following result
\begin{cor}
\label{TogF}
Let $F$ be a fence. Let $w,w'$ be any two Coxeter elements of $\cTh_F$ and let $W, W'$ be the respective groups they generate.  Then any linear combination of indicator functions $\chih_y$ for $y\in S$ is $d$-mesic or homometric under the action of $W$ if and only if it is $d$-mesic or homometric, respectively, under the action of $W'$.\hqed
\end{cor}





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Conjectures and an open problem}
\label{cop}

We now state some conjectures and an open problem; more details appear in~\cite[]{EPRS:rf}. 
Call a sequence of real numbers $a_1,a_2,\ldots,a_n$ {\em palindromic} if 
$
a_i=a_{n+1-i}
$
for all $1\le i\le n$.  
Any $\al$-tiling with $s$ rows has an associated {\em black tile sequence} $b_1,b_2,\ldots,b_s$ and an associated {\em red tile sequence} $r_1,r_2,\ldots,r_{s-1}$.


\begin{question}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$ be palindromic and $F=\uF(\al)$.  Find necessary and/or sufficient conditions on $\al$ for the black sequence   or the red tile sequence  to be palindromic for all rowmotion orbits.
\end{question}
% From the proofs of Theorems~\ref{uF(a,b)}, \ref{uF(a,b,a)}, and~\ref{uF(a,a,a,a)}, we see that the black and red tile sequences are palindromic for palindromic $\al$ with  $s\le 3$, as well as for $\uF(a^4)$ where $a^s$ denotes $a$ repeated $s$ times.
% We have checked by computer that $\uF(a^s)$ has palindromic tile sequences for all $a+s\le 12$ {\em except} $a=4$ and $s=8$, where the orbit containing the antichain $\{s_1,s_{2,1}\}$ (equivalently, $\{x_1,x_7\}$ indexing the entries from left to right), has black tile sequence
% $$
% 21, 20, 18, 18, 19, 18, 19, 21
% $$
% and red tile sequence
% $$
% 5, 4, 13, 4, 9, 8, 5.
% $$
% The black and red tile sequences are related for palindromic $\al$, as shown in the next result.  
For palindromic $\al$, we have the following relationship between the sequences. 
\begin{prop}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$ be palindromic where $\al_i\ge2$ for all $i\in[s]$, and let $F=\uF(\al)$.  Then for any orbit $\cO$ of $F$,
the black tile sequence  $b_1,b_2,\ldots,b_s$  is palindromic if and only if the red tile sequence
$r_1,r_2,\ldots,r_{s-1}$ is palindromic.
\end{prop}
% \begin{proof}
%  We will do induction on $i$.  Because all $\al_i\ge2$, Lemma~\ref{stats} (a) implies that  $\#\cO=b_i\al_i+r_i+r_{i-1}$ for  any $i\in[s]$.  Using the fact that $r_0=r_s=0$ we obtain
% $$
% b_1\al_1+r_1 = \#\cO = b_s\al_s + r_{s-1}.
% $$
% Since $\al_1=\al_s$, This can be rewritten as
% $$
% \al_1(b_1-b_s)=r_{s-1}-r_1.
% $$
% It follows that $b_1=b_s$ if and only if $r_1=r_{s-1}$ which is the base case for our induction.

% The induction step is similar.  We have
% $$
% b_i\al_i+r_i+r_{i-1}=\#\cO=b_{s-i+1}\al_{s-i+1}+r_{s-i+1}+r_{s-i}.
% $$
% By induction, we can assume $r_{i-1}=r_{s-i+1}$.  And $\al_i=\al_{s-i+1}$, so
% $$
% \al_i(b_i-b_{s-i+1})=r_{s-i}-r_i.
% $$
% Thus $b_i=b_{s-i+1}$ is equivalent to $r_{s-i}=r_i$, which finishes the proof.
% \end{proof}

% The condition that  $\al_i\ge2$ for all $i$ in this proposition is necessary.  For example, the proof of Theorem~\ref{uF(a,1,a,1,a)} (b) shows that the black tile sequence of an orbit is always palindromic, but that the red tile sequence is not necessarily so.
% Having palindromic sequences yields certain homomesies for free. 
One reason to care about the palindromic case is the following result.
\begin{prop}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$  where $\al_i\ge2$ for all $i\in[s]$. Also let $F=\uF(\al)$ and $n=\#F$. 
If $\al$ as well as the black and red tile sequences are all palindromic, then one has the following homomesies.
\begin{enumerate}
    \item[(a)] For all $k\in[n]$ the statistic $\chi_k-\chi_{n-k+1}$ is $0$-mesic.
    \item[(b)] If $s$ is odd, then for all $k\in[n]$ the statistic $\chih_k+\chih_{n-k+1}$ is $1$-mesic.
\end{enumerate} 
\end{prop}
% \begin{proof}
% (a) 
% First suppose $x_k\in\uS_i$ for some $i$.  Since $\al$ is palindromic, it follows that $x_{n-k+1}\in\uS_{s-i+1}$.  Now using Lemma~\ref{stats} and the fact that the black tile sequence is palindromic, we see   that
% $$
% \chi_k(\cO) =b_i = b_{s-i+1} = \chi_{n-k+1}(\cO)
% $$
% for all orbits $\cO$.
% This proves the homomesy for our choice of $x_k$. The proof when $x_k=s_i$ for some $i$ is similar using  the palindromicity of the red tile sequence.  The details are omitted.

% \medskip

% (b)  If $x_k=s_i$ for some $i$, then this follows from our assumptions and Theorem~\ref{homo} (d).  So assume that $x_k=s_{i,j}$ for some $i,j$.  There are two similar cases depending on whether $i$ is even or odd, so we will only give details for the former.  From Lemma~\ref{stats} (c), for any orbit $\cO$ we have
% $$
% \chih_k(\cO) = b_i(\al_i-j) +r_{i-1}.
% $$
% Since $x_k$ is an unshared element on $S_i$, we have that $x_{n-k+1}$ is an unshared element on $S_{s-i+1}$.
% And since $x_k$ is the $j$th unshared element from the bottom, we have that $x_{n-k+1}$ is the $j$th unshared element from the top.  So counting from the bottom of the unshared elements in  $S_{s-i+1}$  we have that $x_{n-k+1}$ is element number
% $\al_{s-i+1}-j$.
% Using Lemma~\ref{stats} (c) again, as well as the assumption that the $b_i$ and $r_i$ sequences are palindromic, gives
% $$
% \chih_{n-k+1}(\cO)=
% b_{s-i+1}[\al_{s-i+1}-(\al_{s-i+1}-j)]+r_{(s-i+1)-1}
% =jb_i+r_i.
% $$
% Finally
% $$
% \chih_k(\cO)+\chih_{n-k+1}(\cO)
% = b_i(\al_i-j) +r_{i-1} +jb_i+r_i = \#\cO
% $$
% by Lemma~\ref{stats} (a).  This completes the proof of (b).
% \end{proof}

We conjecture that even more is  true for constant $\al$. 
\begin{conj}
Let $\al=(a^s)$ and let $F=\uF(\al)$. 
\begin{enumerate}
    \item[(a)] The statistic $\chi$ is homometric.
    \item[(b)] If $s$ is odd then the statistic $\chih$ is $n/2$-mesic where $n=\#F$.
\end{enumerate}
\end{conj}
% Again, we have evidence for this conjecture from the results in Section~\ref{ffs}, which prove that it holds for $s\le4$.  Furthermore, computer calculations have verified the truth of these statements for $a+s\le12$.
% One might be tempted to try and prove part (b) by showing that, under the given hypotheses, the superorbits and orbits are the same, and then using Theorem~\ref{n/2-mes}. This would mean that for every ideal $I$ we would have $\Ib$ in the same orbit with $I$.  But this is not always true.  For example, if $\al=(2^7)$ and $F=\uF(\al)$, then the ideal $I=\{x_2,x_4,x_7\}$ has 
% $\Ib=\{x_1,x_3,x_5\}$, which is in a different orbit.


We end with a conjecture which is the antichain analogue of Corollary~\ref{TogF}.
\begin{conj}
Let $F$ be a fence. Let $w,w'$ be any two Coxeter elements of the group of antichain toggles $\cT$, and let $W, W'$ be the respective groups they generate.  Then any linear combination of indicator functions $\chi_y$ for $y\in S$ is $d$-mesic or homometric under the action of $W$ if and only if it is $d$-mesic or homometric, respectively, under the action of $W'$.
\end{conj}


\vspace{-20pt}

\acknowledgements{\vspace{-5pt} We would like to thank BIRS (the Banff International Research Station) for hosting the conference during which the research for this paper began.}

\vspace{-15pt}

%% if you use biblatex then this generates the bibliography
%% if you use some other method then remove this and do it your own way
\printbibliography


%\bibliographystyle{plain}

%\nocite{*}
%\bibliographystyle{abbrvnat}

%\nocite{*}
%\bibliographystyle{alpha}

% 



\end{document}
